Determinants of Employment in the Formal and Informal Sectors of the Urban Areas of Kenya

By

Wambui R. Wamuthenya Institute of Social Studies The Hague, Netherlands

AERC Research Paper 194 African Economic Research Consortium, Nairobi April 2010 **THIS RESEARCH STUDY** was supported by a grant from the African Economic Research Consortium. The findings, opinions and recommendations are those of the author, however, and do not necessarily reflect the views of the Consortium, its individual members or the AERC Secretariat.

- Published by: The African Economic Research Consortium P.O. Box 62882 - City Square Nairobi 00200, Kenya
- Printed by: Modern Lithographic (K) Ltd P.O. Box 52810 - City Square Nairobi 00200, Kenya

ISBN 9966-778-64-0

© 2010, African Economic Research Consortium.

Contents

List o Abstr Ackne	f tables act owledgements	
1.	Introduction	1
2.	An overview of the literature	6
3.	Conceptual framework and methodology	7
4.	Data and descriptive statistics	11
5.	Results and discussion	14
6.	Summary and conclusion	18
Notes	3	22
Refer	ences	28
Appe	ndix A:Variable descriptions and characteristics	31
Appe	ndix B:Results	41

List of tables

1. 2	Total recorded employment excluding small-scale agriculture	3
Ζ.	percentage of relevant population group	3
A1.	Explanatory variables	31
A2.	Total sample	32
A3.	Males – Total sample	33
A4.	Females total sample	34
A5.	Total sample 15–34 age cohort	35
A6.	Total sample 35–64 age cohort	36
A7.	Males 15–34 age cohort	37
A8.	Males 35–64 age cohort	38
A9.	Females 15–34 age cohort	39
A10.	Females 35–64 age cohort	40
B1.	Total sample	41
B2.	Total sample 15–34 age cohort	41
B3.	Total sample 35–64 age cohort	42
B4.	Males total sample	42
B5.	Males 15–34 age cohort	43
B6.	Males 35–64 age cohort	43
B7a.	Females total sample without age-marital status interaction term	44
B7b.	Females total sample controlling for age-marital status interaction term	44
B8a.	Females 15–34 age cohort without age-marital status interaction term	45
B8b.	Females 15–34 age cohort without age-marital status interaction term	45
B9.	Female	46

Abstract

By applying a multinomial logit model and economic theory to labour force survey data, this study examines the determinants of formal and informal sector employment in the urban areas of Kenya. The findings show that the determinants of employment in public, private and informal sectors of Kenya's urban labour market vary by age cohort and gender. Special emphasis is placed on the importance of sex (being male rather than female), marital status, household-headship and education variables, of which the first three illustrate the disadvantaged position of women in the labour market. Education has the strongest impact on formal sector employment, yet most women work in the informal sector despite significant improvements in their education attainment.

Two observations merit concern, high youth unemployment and gender imbalance in access to employment. Unemployment is particularly high amongst women, especially younger women. Younger women are either unemployed or employed in the inferior informal sector (in the sense of low income, precarious and unregulated forms of employment), as opposed to males in a similar age bracket who are likely to work in the private sector.

Overall results confirm that the urban labour market is heterogeneous and reveal how labour supply factors are valued in the labour market. They also indicate the existence of sex discrimination in the labour market. The study raises the following questions for further research while identifying education and employment policy gaps: What specific skills or qualities do employers look for when recruiting new employees? Are the recruitment practices gender balanced? Which training and skills are sought for what sectors? Is the current education system demand or supply driven, and does it equip graduates with adequate skills to become self-employed? Does the current policy environment and infrastructure encourage self-employment? What are the real constraints faced by women in finding reasonable work given their remarkably high unemployment rates? Answers to these questions have broad policy implications towards an achievement of gender balance in education, the labour market and poverty eradication.

Acknowledgements

I would like to convey my profound appreciation to the African Economic Research Consortium for funding this research and for technical training in data analysis and Stata. I am also grateful to the resource persons in particular Prof. Eric Thorbecke, Prof. John Strauss, Prof Anne Bigsten, Prof. Susan Horton, Prof. David E. Sahns and Prof. Steven D. Younger for their various useful comments and suggestions. Lastly, reviews and suggestions by two AERC anonymous referees contributed greatly to improve the research report. I would like to thank them dearly, but all remaining errors are mine alone.

1. Introduction

U rban labour markets in developing countries are widely recognized as having two distinct sectors/segments: regulated/protected formal sector¹ and an unregulated/unprotected informal sector (Pradhan and van Soest, 1995).² Mazumdar (1989) describes an urban labour market structure in a typical developing country as being subdivided into three main categories: the formal sector (public and private), the informal sector³ (comprising the informal sector wage labour, self-employed, paid domestic workers, those earning a monthly salary or those working on casual basis) and the unemployed. This categorization ignores unpaid workers (people who work without pay in an economic enterprise operated by a related person⁴), who form a significant proportion of urban and rural labour force.⁵

Among the most important challenges facing governments in developing countries, including Kenya, is the task of identifying development strategies that can generate new employment and income opportunities, and reduce under-employment and unemployment (Republic of Kenya, 1985, 1986). The urgent need to create employment opportunities is underscored by the higher rate of labour force growth than population growth. Persistent slow economic growth, particularly in the public sector, has forced many people, notably school leavers and college graduates, into marginal activities in small-scale agriculture and the urban informal economy. In recent years, many economies in both developed and developing countries have experienced transformations in their labour market structures resulting from such trends as globalization and economic restructuring.

Since the early 1980s, Kenya has witnessed structural adjustment reforms (structural adjustment programmes – SAPs⁶) that have been accompanied by changes in the structure of employment, incomes and poverty. The economy experienced rapid public sector employment growth especially in the period between 1970 and 1980. This was accompanied by poor job creation in the private sector. The trend took an opposite turn in the period thereafter, which was consistent with the limits of fiscal spending and productive employment within the civil service pursued within the SAPs. Public sector employment was expected to decline in the short run with the implementation of the civil service reform programme, while modern sector job creation would depend heavily on the private sector and self-employment. The period from 1980 to the 1990s can be regarded as the economic reform period and was characterized by more commitment and rigorous implementation of SAPs by the government. The era witnessed measures undertaken to enhance economic efficiency by strengthening the role of the private sector, dismantling price controls, reducing or removing import controls, implementing

financial reforms, and privatizing parastatals. SAPs were adopted with the idea that the accompanying measures would spur economic growth and rectify the worsening macroeconomic imbalance.

The context of this study, then, is one in which the size of the labour force in Kenya has been growing much faster than the rate of growth of formal sector jobs and the fact that both private and informal sectors are expected to play the lead role in employment creation. Table 1 depicts three categories of total employment excluding agriculture. Following implementation of the SAPs, modern wage employment (formal sector) dropped sharply from 77.5% of total recorded employment (excluding small-scale agriculture) in 1988 to 20.2% in 2007. Over the same period, the share of informal sector employment rose just as sharply, from 20.0% to 79.1%. This is an indication that employment is increasingly becoming informalized.⁷ Note that "employment in the informal sector" and "informal employment" are concepts that are now recognized as different aspects of the "informalization" of employment (Hussmanns, 2004).⁸

Year	Total ("000)	Modern (formal) wage employment (%)	Self-employed and unpaid family workers (%)	Estimated informal sector wage employment (%)	
1988	1,736.3	77.5	2.5	20.0	
1989	1,796.2	76.2	2.5	21.3	
1990	2,395.0	58.8	2.0	39.3	
1991	2,557.1	56.4	2.0	41.6	
1992	2,753.2	53.1	2.0	44.9	
1993	2,997.5	49.2	1.9	48.9	
1994	3,355.1	44.8	1.7	53.8	
1995	3,858.6	40.4	1.6	58.0	
1998	5,096.7	32.9	1.3	65.8	
1999	5,492.6	30.7	1.2	68.1	
2000	5,911.6	32.1	1.1	70.2	
2001	6,409.8	26.3	1.0	72.6	
2002	6,866.8	24.8	1.0	74.2	
2003	7,325.7	23.6	0.9	75.5	
2004	7,800.1	22.6	0.8	76.6	
2005	8,281.7	21.8	0.8	77.4	
2006	8,975.6	20.7	0.7	78.5	
2007	9,450.3	20.2	0.7	79.1	

Table 1: Total recorded employment excluding small-scale agriculture

Source, Various Economic Surveys.

A striking feature of the urban labour market, which is also concurrent with the implementation of the SAPs, is a sharp increase in female labour force participation and unemployment rates (LFPRs) particularly between 1986 and 1998. The rise marked a near-convergence of male and female LFPRs in 1998. Male LFPR increased but only marginally: Urban female LFPR increased from 6.4% in 1977 to 38.8% in 1986 (Republic of Kenya, 1988) and to 85.9% in 1998, while over the same period, urban male LFPR dropped from 83.9% in 1977 to 82.2% in 1986, then rose to 86.9% in 1998 (Republic of

Kenya, 2003). Female unemployment rates, particularly among young females, also rose sharply – from 6.4% in 1977 to 24.1% in 1986 and to 38.1% in 1998. Meanwhile, the male unemployment rate remained more or less constant at 6.8% in 1977, 11.6% in 1986 and 12.5% in 1998, thus a minimal change between 1986 and 1998. Thus the gender gap in employment and the unemployment rate remained wide. On the whole, the urban unemployment rate shot from 6.7% in 1977, to 16.2% in 1986 and to 25.1% in 1998. A greater part of the increase in overall urban unemployment was driven by the tremendous increase in overall female unemployment.

Despite the tremendous increase in the urban female LFPR, both formal and informal sectors remain highly male-dominated (Table 2), while unemployment has remained outstandingly high among women. With depressed formal employment growth (in both public and private sectors) coupled with the slowdown in overall economic activity that has occasioned massive retrenchments in both public and private sectors, the informal sector has provided the greatest opportunity for employment. The effect of civil service reforms combined with the restructuring programmes in the private sector exacerbated the already volatile unemployment situation in the country. It has aggravated competition for the limited number of new jobs, estimated to be just over 100,000 a year, in a situation in which new entrants to the labour force number more than 400,000 annually. In addition, a lack of demand for increasingly educated labour has contributed to increasing unemployment and especially youth unemployment (also caused by rural to urban migration) and growth of informal sector employment, while the effect of declining real wages on family incomes, pushes urban households to attempt to diversify their income sources through additional activities in the informal economy. Manda and Sen (2004) observe that the economic reform period was accompanied by a dramatic increase in informal sector employment, which became gradually more insecure because of the low survival rate of firms in the sector and low pay as compared with the formal sector. Unemployment increased in both rural and urban areas and especially among women in the urban areas. These authors find that the reform period also witnessed a shift in labour demand in favour of highly skilled labour, a decline in permanent fulltime workers, and an increase in part-time and casual workers as a cost cutting strategy.

Table 2:	Employed and unemployed persons by sex and sector as percentage of
	relevant population group, 1988

Sector	Female	Male
Formal	23.4	53.2
Informal	30.7	32.1
Unemployed	45.9	14.8
Total	100.0	100.0

Source: Own computation from the 1998/99 Labour Force Survey (LFS) data

To sum up, Kenya has over the last couple of decades witnessed structural adjustment reforms that have been accompanied by changes in the structure of employment, incomes and poverty. Generally, the Kenyan economy shows a mixed performance with much better performance in the early 1970s.⁹ The situation deteriorated from the 1980s after the implementation of the SAPs coupled with other external and internal shocks.¹⁰ It got

particularly worse between 1990 and 2002. Real GDP growth rate performance has been poor; employment creation has been low; unemployment has increased; and poverty has risen significantly in both urban and rural areas – marked by a substantial greater increase in urban poverty between 1992 and 1997 than in rural poverty, with income inequality increasing over time.¹¹ Additionally, changes in the urban employment structure relate to the contraction of formal employment and an expansion of informal sector employment where earnings are considerably lower¹² and more irregular than in the mainstream sector, thus further worsening the economic situations of households.

In the meantime, the economy witnessed rapid growth of the informal economy, a process accompanied by much greater reliance of households on multiple and diversified formal and informal sources of income – in more specific terms, the reluctant "stop-go" implementation of economic reforms accompanied by poor growth performance led to a protracted process of real earnings decline that adversely affected the livelihoods of households. This in turn led to a significant rise in the incidence of poverty. Over and above, the standard of living has deteriorated, as evident from the poor performance of the per capita GDP, real earnings growth, high inflation and a drastic increase in urban poverty. The decline in real earnings was highest in the periods 1982–1994 and 1991–1994 when there was more rigorous implementation of SAPs.¹³ The period 1991–1993 was Kenya's worst economic performance since independence in 1963.

Alongside these negative macroeconomic developments Kenya has also experienced a rapidly increasing labour force and a marked expansion of education (since 1963).¹⁴ However, the economy has not been able to generate enough jobs to meet the demand.¹⁵ The rapid expansion of education, which has led to an increase in the level of education of the labour force, implies that productivity endowments on which individual participation behaviour depends have changed over time.

This study is motivated by the fact that the Kenyan economy has witnessed a significant decline in modern wage employment's share of total employment and the reverse trend in the share of informal sector employment. These trends call for an analysis of job attainment within the key sectors of Kenya's urban labour market by assessing the importance of human capital, individual and household characteristics on an individual's type of employment. This analysis is relevant because different sectors have different opportunities for skill realization and earnings, while such a study can yield indications of heterogeneity or discrimination in the labour market. Additionally, an understanding of the sectoral determinants of the labour supply has policy implications to do with eradication or minimization gender gaps in employment, human capital investment, and the level and distribution of income.

Against this background, the key issues addressed in the paper are as follows: The Kenyan urban labour market structure, consisting of the three broad categories of formal (modern wage employment – public and private sectors together), informal sector employment and unemployment, implies that some kind of sorting process takes place to situate workers in these categories. Once an individual has made the decision to participate in the labour force, what personal and socioeconomic characteristics are important for this process? More specifically, what labour supply factors are associated with a greater (lesser) likelihood of being in formal (public and private) and informal sectors? Is there a significant variation by gender in job attainment in view of women having attained analogous LFPR to men in 1998? Do the factors differ between younger

and older cohorts?

To embark on these questions, the approach of this paper is as follows: First, as summarized above, it explores changes to the Kenyan labour market brought about by SAPs and asks what characteristics make men and women more inclined to some sectors than others. Second, the paper analyses (both descriptively and empirically) a labour force survey (LFS) of a single point in time (cross-sectional data of 1998) in order to identify labour force characteristics associated with a greater (lesser) likelihood of public, private and informal sector employment. Empirically, this is undertaken using a multinomial logit model (MNL).

The reminder of the paper is structured as follows, Section 2 provides an overview of literature and Section 3 outlines the conceptual framework and methodology for the study including a specification of the MNL model and variables to be estimated. It also highlights the major limitations of the study. Section 4 describes the LFS data. Results are reported in Sections 5 and Section 6 concludes.

2. An overview of the literature

Multiple empirical evidence exists on labour supply factors that affect an individual's labour participation decisions in the context of both developed and developing countries. (See for example, Gong and van Soest, 2000; Hunter and Gray, 1999; Shoshana and Shoshana, 1998; Kolev, 1998; Lanot and Muller, 1997; Manda, 1997; Makonnen, 1993; Blundell et al., 1987; Myint, 1985; Nwanganga, 1980; Ranis and Fei, 1961; Lewis, 1954.) The studies contain a subset of the following social and economic characteristics among the explanatory variables: female wage, male wage, age, schooling (education attainment), marital status, number and age of children, occupation of primary job, region of residence, religion, and non-labour income. From a methodological point of view, most studies reviewed have discussed in detail some of the econometric issues encountered: measurement problems, selectivity bias problems and weaknesses in using an ordinary least squares (OLS) estimation technique. These issues are addressed in subsequent sections of this paper.

In the Kenyan context, the studies that exist (Manda, 1997; Milne et al., 1990, 1994) – all based on the 1986 labour force survey – have analysed the determinants of participation by simply modelling the labour force participation decision into two categories (employed and unemployed) without taking into account that the determinants also vary by employment sector among the employed. A study by Mariara (2003), based on the 1994 Welfare Monitoring Household Survey data, although mainly concerned with issues of gender wage discrimination, has partly looked at the determinants of participation by disaggregating the urban labour market according to formal (public and private) and informal sectors and by sex. However, determinants of employment may vary between younger and older persons. Here, using a more recent data set, this study bridges the gaps by looking at the determinants of employment by gender and age cohorts.

3. Conceptual framework and methodology

D espite its limitations, the theory underlying participation decisions and number of hours worked has typically been modelled within the context of a standard neoclassical microeconomic model, which is the main standard theory widely applied in the empirical analysis of labour supply (Gray et al., 2002, 2003; Manda, 1997; Van den Brink, 1994; Heckman, 1979; Killingworth, 1983; Becker, 1965).¹⁶ Labour force participation (the state of being employed or unemployed) is one dimension of labour supply in that individuals not only make a choice of how many hours to work, but also make a simultaneous decision of whether to work at all.

Within the neoclassical labour supply framework, individuals as rational actors maximize their utility and are willing to enter employment governed by the fundamental requirement that the market wage exceeds the reservation wage.¹⁷ How a variable affects women's employment decisions depends on how it affects the reservation and market wages or both. A variable that increases the reservation wage, such as the presence of young children, availability of non-labour income,¹⁸ changes in tastes and preferences towards leisure and other non-market activities, the level of structural, cultural and incompatibilities between family and work (such as availability of childcare opportunities and attitudes towards working mothers), decreases the probability of being employed. A variable such as education can strengthen women's attachment to the labour market by increasing their potential earnings in the labour market, thus increasing their market wage.

The labour force status of an individual as predicted by the neoclassical analysis is thus determined in a two-stage process. First, an individual decides whether or not to supply labour to the market. Second, whether they are employed or not is determined by a combination of factors including labour demand (employer preferences, such as skills, experience, education, marital status and sex), incentives to search actively for work and to go for any job offers available. Major drawbacks of the theory are that it ignores the interdependence of household members and therefore their decision making, and that it fails to distinguish between productive and recreative activities (Van de Brink, 1994).¹⁹

Most of the empirical effort of this paper is devoted to the supply-side of the labour market with a look at the determinants of employment (sectoral choice as opposed to being unemployed) at household and individual levels. It should be stressed that the supply-side of a labour market is equally dependent on its demand side and vice versa. Labour markets function through an interaction of workers and employers, resulting in patterns of wages, employment and income. As discussed above, participation in the labour force involves a decision by individuals on how to allocate their time, and a decision on the part of the employer to offer the individual a job. It is this interaction of supply and demand that determines whether a person participates in the labour force. In an empirical estimation of labour supply, it is not always possible to specify the demand schedule, especially when the analysis is confined to the household level, because information on labour demand variables is lacking.

Empirical model for the determinants of sectoral choice

This study uses a MNL model²⁰ in which individuals are sorted into four labour force categories, public sector, private sector, informal sector, and unemployed. The model allows the dependent variable to take four mutually exclusive and exhaustive values, j=1, 2, 3 or 4 and is defined as follows,

$$(Yi = j) = \frac{e^{\beta_j X_i}}{\sum_{m=1}^{4} e^{\beta_m X_i}}$$

where:

 $Y_i = 1$ if an individual is employed in the public sector $Y_i = 2$ if an individual is employed in the private sector $Y_i = 3$ if an individual employed/works in the informal sector $Y_i = 4$ if an individual is unemployed X_i represents a specific explanatory variable

Thus, the dependent variable has four categories/outcomes.²¹ In order to facilitate understanding of the effects of the logit coefficients, marginal effects or predicted probabilities (that is, change in predicted probability associated with changes in the explanatory variables²²) are developed on the basis of the MNL model of being in each of the four outcomes. Marginal effects (ME) are evaluated at the sample mean.

An important remark pertaining to informal sector employment has to do with how to treat the various categories of informal sector workers: wage workers, unpaid family workers, working employers and own-account workers. The issue is whether to include or exclude the last three from the analysis. Since the informal sector is now a major employer – currently accounting for over 75% of total employment – it makes sense to include these three categories together with informal sector wage employees. Moreover, as mentioned above, non-agricultural household enterprises are a foundation of the Kenyan economy in which own-account workers and unpaid family members together make up more than 70% of the entire employment.

The independent variables include personal and household characteristics as well as the social economic background. Among personal characteristics are age, level of education, marital status and household-headship, and wages in the market. Household characteristics include childcare responsibilities such as number of young children (below school age), the size of the household and the presence of female relatives in the household. Age is included to pick up the life cycle effects and as a measure of potential labour market experience.²³ Education is captured by years of schooling or highest level of formal education completed.

The structure of the family has an important bearing on women's participation in the labour market and the sectors they go into. Scholarly consensus has been that women are deterred by marriage and children in the context of labour division within households where husbands specialize in market work and are the breadwinners, and women in childrearing and household work. Married women with young children are thus more likely to be unemployed than to be engaged in formal or informal sector employment. But this depends on the level of education of a woman, as women with high education levels are less constrained by the need to care for young children because they can afford to hire domestic help. As long as it is wage employment where there is little flexibility and the hours of work are fixed, women may select themselves into the inferior informal sector. To some extent, the sector somewhat enables them to combine productive and reproductive/ care work.²⁴ In the Kenyan context, the greatest number of informal sector workers amongst women is indeed own-account workers (self-employed).

Most empirical studies find a negative relationship between the number of children in the family and the probability of participation by the wife (Kaufman, 1994). The negative relationship is particularly associated with children of preschool age. This standard gender analysis may breakdown in the Kenyan context where domestic service is affordable among educated women engaged in fulltime jobs or among those engaged in self-employment in the informal sector, which often allows them some flexibility to combine productive and reproductive work.²⁵ Concerning informal sector employment, however, this factor depends on whether one is really able to combine work and caring for children as this may depend on the age and number of children. Since decisions to have children is left out in this paper.

Presence of female relatives in a household would be expected to increase the probability of formal and informal sector employment, as opposed to being unemployed, on the assumption that their presence reduces the burden of caring for children and handling domestic chores for other women in a household thus allowing time to engage in market work. Presence of female relatives is included in the models of married and single women.

It would be expected that being a household head increases the probability of being in employment in either of the two sectors regardless of sex or marital status. Being a household head is expected to reduce the likelihood of being unemployed.

Household size could have either a positive or a negative effect on the probability of being in employment (both sectors). On the one hand, a large household may mean heavier household chores for a woman and therefore a higher reservation wage. In this case, the effect on the probability of being employed in either of the sectors would be negative and positive for being unemployed. On the other hand, a large household may mean an increase in the financial constraints of her household, thus requiring her involvement in the job market. A large household with non-working adult members, especially females, may relieve women of some of the domestic responsibilities such as looking after young children

thus enabling them to take up market work. Here, the effect on the probability of being in employment would be positive and negative for being unemployed.

It should be noted that there is yet another possible endogeneity between the endogenous variable and the exogenous variables (the left-hand side and the right-hand side variables). For instance, marital status, female relatives and household-head variables could be a source of this bias. Specifically, labour market participation in a particular sector may also determine who is regarded as the household head. More female relatives might come to live with a relative who has a stable job in whatever sector. Similarly, marital status may also depend on the job – persons without jobs may not have the resources to marry or stay married. This should be kept in mind while interpreting these variables.

Partner's income and partner's employment status are known to influence the labour supply decisions of women although the effect is not obvious. If partner's income is taken as exogenous under the assumption that the decision to participate is not simultaneously decided within a couple, it would be expected that a higher partner income reduces the probability of participation in both sectors. In the context of worsening economic circumstances, women may be more inclined to work or search for work in order to augment the partner's falling real income (hence an inverse relationship between the likelihood of being employed and partner's income) or to cushion against the partner's potential job loss resulting from employment restructuring. Women (especially the highly educated) may be more induced to work for other reasons independent of partner's earnings such as the need for self-security or economic empowerment. Women whose partners are unemployed may become more inclined to work because they have further economic needs. On a different perspective, if men and women are affected by same economic shocks from the labour market, it may mean that if a man is unemployed, a woman will be more likely to be unemployed as well. Partner's real income and education are included in the model of married women.

On wages in the market, recall that the theory underlying the participation decision stems from a standard neoclassical microeconomic model in which an individual's decision to work is determined by the difference between the market wage and the reservation wage. Since by definition individuals who do not earn a wage and are yet part of the labour force – for instance the unemployed do not have observed wages – all that could be assumed about them is that their market wage is below their reservation wage. More specifically, in the analysis of the labour supply behaviour especially for women, this assumption is inappropriate, partly because many women are unemployed while the wages for non-workers are unobserved leading to econometric problems to do with model specification, measurement and biased estimates.²⁶ Ways to avoid these problems would be to estimate the employment choice equation in a reduced form where both market and reservation wages are replaced by their determinants or by using a direct measure of the market wage. In the former case, it means that the wage levels are not included among the independent variables in the model. This paper follows this method. In the latter case, since unemployed and inactive persons have unobserved wages, the variable used is the predicted wage. Appendix Table A1 provides a list of variables to be estimated and details of their construction.

4. Data and descriptive statistics

The Central Bureau of Statistics (now the Kenya National Bureau of Statistics) of the Ministry of Finance and Planning in Kenya has collected labour force survey data (cross-sectional data) at different points in time: 1977, 1986 and 1998. The study uses the 1998 data. The 1998/99 Integrated Labour Force Survey, conducted in all administrative districts of Kenya constituted in 1989, utilized the 1989 National Sample Survey and Evaluation Programme (NASSEP) frame created under the 1989 populationhousing census. The framework followed a two-stage cluster design of 1,139 clusters, out of which 209 were urban; in these 1,938 households were interviewed. In total, 6,646 individuals were interviewed.

The analysis focuses on individuals in the labour force²⁷ age (15–64), excluding the inactive population, young children, fulltime students, the retired and the sick/handicapped. From the original sample of 6,646 observations, 83 observations with incomplete information were deleted, leaving 6,563 observations in which 4,008 are in the labour force age. Of these, 2,557 are employed (formal and informal sectors), 960 are unemployed and 491 are inactive. However, the sample for the analysis consists of 3,238 observations (the employed and the unemployed together, 1,547 females and 1,691 males) of which 537 are public sector employees, 724 are private sector employees and 1,017 are informal sector employees. The remaining 960 are unemployed. Some respondents may hold multiple jobs, but the data do not contain information on this issue. Information on the sample characteristics is provided in Appendix A.

As shown in Table A3 (entire sample), the average age is the highest among public sector workers. The proportion of household heads is higher among public and private sector workers (about 78% and 77%, respectively, compared with 61% and 26% for informal sector workers, and the unemployed respectively). More males than females in the labour force are likely to be engaged in all sectors (about 67%, 74% and 53% for the public, private and informal sectors, respectively). However, unemployment is highest among females (about 74%). Most members of the labour force are married.

Persons engaged in the public and private sector are more likely to have secondary level education (76% and 59%, respectively). Most of those engaged in the informal sector and the unemployed have primary level education (47% and 45%, respectively), although the proportion of those with secondary level education is substantial (about 41% and 43%). The proportion of persons with primary level education is lowest in the public sector (about 15% compared with 32% for the private sector). The proportion of persons with university level education, although quite small, is highest among public sector employees (8% compared with 5% for the private sector and 2% each for the

informal sector and the unemployed). Greater shares of people with no education are found among the unemployed (about 11%) and the informal sector (9%), compared with only 1% and 4% for the public and private sectors, respectively.

A similar analysis by sex (Tables A3 and A4) shows that the mean age for males in the labour force is 38, 34 and 30 for the public sector, for the private and informal sectors each, and for the unemployed, respectively. The mean age for females is lower by four years for the public and private sectors each, and by two years each for the informal sector and the unemployed. Thus most males in the labour force are older than females. Most married females are unemployed, followed by public and informal sector employees, while most married males are public sector employees, followed by private and informal sector employees. The proportions of male and female household heads are highest in the public sector.

Most male and female employment in the public and private sectors comprises those with secondary level education (87% and 55% among females and 70% and 61% among males for the public and private sectors in that order), while low levels of education characterize informal sector employment and unemployment. A snapshot of the general education trends of women in Kenya shows the following: Although school enrolment rates have risen, most girls do not progress to higher levels after completing primary level education. Some tend to get married or enter the labour force early and take up low earning occupations or income-generating activities. A significant number of those who progress to the secondary level drop out. Those who succeed in completing this level hardly proceed further. Progressively, more dropout trends among women partly contribute to their increasing unemployment rates, a scenario further exacerbated by their increasing rural to urban migration.²⁸

Wages in the public sector are the highest followed by those in the modern private sector.²⁹ Earnings of women are far below those of men.³⁰ The disparity between the wages from the modern sector and the rest of the economy is large – average wages in the formal sector are about twice those in the informal sector and about four times those in the small-farm sector.³¹ The modern sector hires more educated workers, hence their wages tend to be higher than those of workers in the rest of the sectors. Disparities in income (referring more directly to disparities in labour earnings) across Kenyan households derive to a degree from disparities in access to work. Persistently low household income reflects the incidence of low-productivity informal and traditional farm/pastoral activities. Only a small proportion of working Kenyans (specifically paid employees in modern wage employment) obtain adequate earnings from labour activities (Pollin et al., 2007).

As can also be observed in Table A3, the distribution of males in the labour force according to the four labour force categories is as follows: about 21%, 32%, 32% and 15% for the public, private, informal sectors and the unemployed, correspondingly. Among females (Table A4), the distribution is 11%, 12%, 31% and 46%, respectively. Thus the majority of males in the labour force (53%) are engaged in formal sector employment. A majority among females are not employed at all (45.7%) followed by those engaged in the informal sector – the proportion among females is relatively slightly higher in the informal sector (31%) compared with the formal sector (23%). Thus, women are predominantly engaged in the informal sector where education and skill levels are low or moderate. The formal sector where education and skill levels are high is male-

dominated; as noted, women constitute slightly less than a third of formal sector employment in Kenya.

13

In sum, most women work in the informal sector and are mainly concentrated in precarious employment activities. Unemployment has remained outstandingly high among them and is also characterized by low education levels.

So far, the descriptive analysis gives a general picture of the average characteristics of the entire urban labour force and of males and females separately. Descriptive statistics for younger and older persons are presented in tables A5–A10. As already highlighted, a majority of the Kenyan urban labour force is likely to be employed in the private and informal sectors. This is consistent with the changes and polices pursued by the Kenyan government of restructuring public sector employment to make it more efficient in creating an enabling environment for the private and informal sectors, which are in turn expected to create jobs for the ever-increasing labour force in the long run. However, there are quite significant differences in the four states – public sector employment, private sector employment, informal sector employment and the unemployed – when the analysis is broken down by sex (as discussed above) and by age-cohorts (as follows). The scenario is particularly astounding among the unemployed and younger persons (aged 15–34).

The distribution of all younger persons (Table A5) is as follows: 12%, 23%, 30% and 37% for the public, private and informal sector and the unemployed, respectively. For the older cohort (Table A6), the distributions over the same categories are 25%, 23%, 34% and 19%. Thus younger persons are likely to be in the unemployed category (as in many parts of the world) followed by the informal sector category, while the majority among the older cohort is employed in the modern sector.

The key point to note is that unemployment is particularly serious among the youth. This is partly attributable to the increasingly hard challenge the Kenyan economy has experienced over the years of trying to create employment for the ever-growing numbers of the labour force (mainly comprising young persons). The situation has been exacerbated by a fertility increase (although currently at a declining trend) coupled with rural to urban migration. This also partly explains the growth in informal sector employment as a steppingstone for unemployed job seekers.

By sex, the distributions among younger males (Table A7) are 15% for the public sector, 33% each for the private and informal sectors, and 19.1% for the unemployed. For older males (Table A8), the distributions are 29% for the public sector, 31% each for the private and informal sectors, and 10% for the unemployed. Thus, although unemployment is relatively lower among males than among females, it remains a youth problem even among males.

Among younger females (Table A9), the distributions are 9%, 13%, 28% and 51% for the public, private and informal sectors, and the unemployed, respectively. For older females (Table A10), the same distributions are 18%, 11%, 38% and 34%. Thus, while unemployment is also high in the older cohort (unlike in a similar male cohort), it is more serious among the younger cohort. Unemployment is clearly more serious among females. As observed above, the majority of younger males are likely to be private or informal sector workers. This triggers a concern as to why a majority of females in the younger cohort are unemployed, and if employed, in the inferior informal sector.

5. Results and discussion

E stimates of the probability of sectoral choice are presented in Appendix B. There are clear cohort and gender differences in the choice of sector. In the interest of brevity, emphasis is placed on the importance of sex (being male as opposed to being female), marital status, household headship and education.

For the entire urban sample (Table B1), sex (positive coefficient) is important for all sectors. Marital status enhances the likelihood of public sector employment only. All the three levels of education are important (with a positive effect) for both public and private sector employment, while only primary level education is important for informal sector employment. Household-headship is important for all sectors.

Although some of the determinants of employment appear similar across the three sectors, their marginal effects (MEs) differ and are interpretable as follows: A positive (negative) sign implies that a variable increases (reduces) the probability of a particular outcome. For instance, being a household head (as opposed to not being a household head) increases the likelihood of working in the public, private and informal sectors with MEs of about 7%, 15% and 14%, respectively, for an obvious reason. As expected, being a household head decreases the probability of being unemployed by about 36%.

The sex variable has a positive coefficient and is statistically significant in the three sectors. Its MEs is negative in the public and informal sectors (by just about 1% and 3%, respectively) and in the unemployed outcome (by about 12%), but positive in the private sector (by about 15%). This implies that being male as opposed to being female enhances the likelihood of being in private sector employment and reduces that of being in the public and informal employment and of being unemployed. Although this is subject to further research, it may connote some kind of sex discrimination in the labour market in which private sector employers prefer to hire men rather than women despite both groups having probably attained the same level of human capital. From a non-discriminatory point of view, employers may have a preference for men to women because of their varying levels of human capital skills, in which men in general are more advantaged than women. Still, as long as it is wage employment where there is little flexibility and hours of work are fixed, women may select themselves into the inferior informal sector in order to also cope with care work (caring for children and domestic chores). To an extent, this sector enables them to combine productive work and care work.³²

The coefficient and MEs of the marital status variable are positive and significant in the public sector relative to being unemployed, but negative and insignificant in the private and informal sectors. It could be that married people prefer to work in the public sector because working conditions are better despite low pay, as compared with the private sector, or that persons join the public sector when they are fresh from college and while still single but as they get married later in their lives and as jobs become increasingly difficult to find in the private sector, they get stuck in the public sector. It could also be that married persons may prefer to work and remain in the public sector despite lower wages (in comparison with wages in the private sector) because of greater flexibility (ability to simultaneously hold multiple income-generating activities in the informal sector), guaranteed benefits (sick and maternity leaves) and better working conditions (fixed/ stable employment contracts).

The positive effect of education implies that it enhances the opportunities for working relative to having no education (the omitted category). This chance increases with the level of education and is therefore highest for university level education. In the public sector, the ME of education are about 11%, 25% and 51% for the primary, secondary and university levels, respectively. In the private sector, they are about 7%, 12% and 0.3%, respectively. In the informal sector and among the unemployed, the MEs are negative – thus diminishing the likelihood of being in these two outcomes – by about 5%, 20%, and 31% for the primary, secondary and university levels, respectively, as pertains to the informal sector and by about 13%, 18% and 20% for the unemployed. Thus, the positive effect of education on enhancing employment is highest in the public sector and strongest for university level education. In the private sector, secondary level education has the strongest effect.

Estimates for younger and older cohorts (without separate samples by sex) are in tables B2 and B3, respectively. The effect of education also varies by cohorts and by sector. For instance, among younger persons, university and secondary levels of education enhance the likelihood of public sector employment, with university education having the strongest ME of 36%, compared with 9% for secondary level education. For the private and informal sectors, both primary (positive coefficient and MEs of 8% and 14% for the private and informal sectors, respectively) and secondary levels (positive coefficient and MEs of 16% and negative 1%, respectively, for the private and informal sectors) are significant for the likelihood of employment, while university education appears unimportant. Nevertheless, the younger cohort's secondary level education has the strongest impact in the private sector, while primary level education has the strongest impact in the informal sector.

Among the older cohort (age 35–64), all three levels of education augment participation in the public sector (positive coefficients with ME of 30%, 46% and 57% for primary, secondary and university levels, respectively), while in the private sector only secondary and university levels of education are important (positive coefficients with a positive ME of 3% for the secondary level and -8.2% for the university level). Thus university education has the strongest impact on the probability of participating in public sector employment for both age cohorts. The highest positive effect of education (all three levels) is observed among the older cohort engaged in the public sector. The highest positive effect of education in the private sector is observed among younger persons, especially that of secondary level education. Primary and secondary levels of education are important for informal sector employment among the younger cohort, with primary level education having the strongest effect. All three levels of education are unimportant for informal sector employment among the older cohort.

In both cohorts, sex (being male rather than female) increases participation in private sector employment. Sex is also important among younger persons as concerns informal sector employment.

For the younger cohort, marital status does not affect the likelihood of employment in all sectors. After interacting it with age, however, the variable becomes negative and significant in all sectors. The interaction term is positive and also significant, connoting that the effect of age on married younger persons is more positive than its effect on single younger persons in all sectors relative to being unemployed.³³ These results imply that for persons in this cohort, being married discourages participation in all sectors but an increase by a unit in age for married persons enhances it. In reality, this is perhaps due to an increased financial burden of raising children or it could be that the reproductive burden hinders younger persons from working. For the older cohort, marital status is unimportant except in the public sector where the relationship is positive while the age-marital status interaction term is insignificant in all sectors³⁴ – for this cohort, the reproductive burden may not be a hindrance as children are either at school fulltime or are independent.

By sex, Table B4 shows that for males (without age-cohort breakdown), marital status has a positive correlation with sectoral choice (all sectors) with the strongest predicted probability being in the public sector. This result is not surprising and arises because most married males are also the household heads (main economic caregivers) and therefore being married necessarily enhances their likelihood for employment irrespective of sector. Moreover, the Kenyan family system is a highly patriarchal one in which, culturally, men are household heads in terms of key decision making and providing main economic support for the household. Household headship has a positive coefficient (MEs of 6%, 13%, and 8% for public, private and informal sectors, respectively). All three levels of education augment their chances for securing a public sector job, with university level education having the strongest effect. Both secondary and university levels boost male participation in private sector employment. However, despite positive correlation of these two levels with participation in private sector employment, their marginal effects are statistically insignificant. Primary level education is important for informal sector employment.

By age cohorts, household-headship and marital status are important for all the three sectors among younger males (Table B5). All other factors are insignificant. The strongest positive effect of household headship is in the private sector (about 29% compared with 5% of the public sector and 6% of the informal sector). The same holds for marital status – about 9% compared with 5.1% of the public sector. The MEs of marital status are negative in the informal sector (about 1%) and in the unemployed outcome (about 14%), thus reducing young men's likelihood of being in these two outcomes. Of importance to note about this cohort is that none of the three education levels is significant in any of the sectors. For the older cohort (Table B6), household headship and marital status and all three levels of education determine participation in public sector employment, while only secondary and university levels of education are important for private sector employment. The unique points to note about this cohort are that household headship and marital status seem to have no effect on participation in the private sector and that none of the variables are significant for informal sector employment. The age–marital status interaction term is insignificant in both cohorts.

Among females (Table B7a), household-headship is important for all sectors – positive coefficient and positive MEs of 4%, 5% and 26% for public, private and informal sectors,

respectively). Secondary and university levels of education have a positive influence on public sector employment (with MEs of 19% each), while all three levels of education are important for private sector employment, particularly university level education (MEs are 11%, 14% and 42% for primary, secondary and university levels, respectively) – the MEs are remarkable for females in this sector. Marital status³⁵ has a negative effect on private sector employment, thus increasing women's reservation wage while female relatives are also important for public sector employment with a positive effect, thus reducing their reservation wage. The age–marital status interaction term is significant in the private and informal sectors (Table B7b) unlike in the male sample. Education is unimportant for the informal sector, thus reducing the reservation wage.

For younger females, the variables (without age–marital status interaction term, as presented in Table B8a), household-headship is important for all sectors (with MEs of 3%, 6% and 28% for public, private and informal sectors, respectively). Being married reduces the chance for private sector employment. The presence of female relatives in a household increases the likelihood of public sector employment while that of older children of school-going age is positive in the informal sector. Secondary level education improves young women's odds of employment in the three sectors with MEs of 9%, 14% and 6% for public, private and informal sectors, respectively. Primary level education is important only for informal sector employment (MEs of 15%), while university level education is relevant only for private sector employment (MEs of 36%). The age–marital status interaction term (Table B8b) is positive and significant in all the sectors.

Determinants of employment among older women are summarized in Table B9. Being a household head increases the likelihood of being employed in all sectors, with MEs of 7%, 5% and 26% for public, private and informal sectors respectively. Only secondary and university levels of education are important for public and private employment with MEs of 41% and 26%, respectively, for the public sector and 9% and 39% for the private sector. The presence of older children is positive and important for informal sector employment, thus increasing the market wage. Household size is important for public sector employment – and negatively signed – thus increasing the reservation wage. Marital status for this cohort has no effect on participation in all sectors, while the age–marital status interaction term is insignificant.

6. Summary and conclusion

The paper assessed the determinants of formal and informal sector employment in the urban areas of Kenya based on labour force survey data of 1998. Descriptive statistics show that a majority of the Kenyan urban labour force is likely to be employed in the private and informal sectors. This is consistent with the changes and polices pursued by the Kenyan government of restructuring public sector employment to make it more efficient in creating an enabling environment for the two sectors which are in turn expected to create jobs for the ever-increasing labour force in the long run. However, there are quite significant differences in public, private and informal sectors of employment and the unemployed outcomes when the analysis is broken down by sex and by age cohort.

Of particular concern are the unemployed and informal sector categories, with special attention to gender imbalance in access to employment. Unemployment is clearly a youth problem and its incidence is more serious among women (younger women especially) than men. Formal sector employment in both public and private sectors is male-dominated while women occupy the inferior informal sector (inferior in the sense of low incomes, precarious tenure and unregulated forms of employment) or are unemployed. As matter of fact, a near majority among females are unemployed (46% compared with only 15% of males) followed by those engaged in the informal sector. This phenomenon persists, despite women becoming increasingly well-educated.

The findings reveal that the key labour supply factors important for employment in the public, private and informal sectors of the Kenyan urban labour market vary by cohort (younger and older persons) and by sex. What may appear as the effect of each factor could be misleading, if an analysis failed to take into account this distinction. This variation reflects how labour supply factors are rewarded in the labour market and corroborates heterogeneity in Kenya's urban labour market. The results also give indications of discrimination in the labour market. Special emphasis is placed on the importance of the variables *sex, marital status, household headship* and *education,* of which the first three reveal the disadvantaged position of women in the labour market.

In terms of the *sex* of respondents, results indicate that being male rather than female enhances the likelihood of employment in the three sectors, particularly the private sector. This means that men have a higher chance than women of working in the three sectors as opposed to being unemployed. Of policy concern and scope for further research is to identify whether this discrepancy is a result of labour market discrimination against women (identified in the literature as sex discrimination) or is justified on the basis of human capital skills. As for *marital status*, being married as opposed to being single reduces women's (especially younger women) chance of employment in the private and informal sectors. It could be that there are other underlying factors that inhibit married women from working such as sex discrimination and care work (where women may voluntarily or involuntarily decide not to work). This further illustrates the underprivileged position of women in the labour market and it requires further attention.

19

Household headship is also important. Household heads are more likely to work than non-household heads for obvious reasons. The difference lies in which sectors they are likely to work when the analysis is done for males and females separately. Female household heads are more likely to work in the informal sector irrespective of age, while male household heads are more likely to work in modern wage employment. This has implications for poverty and in particular its incidence in female-headed households.

The study identifies *education* as a major factor determining participation in modern wage employment. Of policy concern is that a high level of youth unemployment persists, despite a significant improvement in education attainment. Related to this is whether the youth are equipped with skills competitive enough for a changing labour market or at least adequate to enable them to create their own employment. Moreover, against the background of currently very high urban unemployment rates among young graduates and first-time job seekers, informal businesses that are precarious in nature emerge among the few potential options of becoming economically active. However, if the environment is not encouraging for young persons with human capital skills and good business ideas , it is apparent that new technology schemes will be rapidly over and done in support for survival schemes within the typical service (retail trade) industry as is the case among non-agricultural household enterprises.

Worth mentioning is that education appears to have no impact on informal sector employment, although primary level education does have a minimal effect among males and females in the younger cohort. Possible reasons are that informal sector employment requires skills and capital, rather than a high education level as would be the case for modern wage employment. The study casts doubt on the extent to which the informal sector is able to absorb the excess and growing number of young, educated and unskilled workers, given a significant decline in the formal sector employment. This has implications for the rising levels of urban poverty, crime and potential for civil disorder in Kenya.

It is commonly acknowledged that the majority among the poor in Kenya, as in other sub-Saharan African economies, have low education levels and derive their incomes from the informal sector. Poverty is particularly high among women and children and, as noted, most women derive an income from the informal sector. Linked to this is a potential problem of life-course and intergenerational poverty transmissions – conditions of childhood that can lead to poverty throughout the life-course and affect transfers of poverty to the next generation. Escape from poverty depends on numerous factors including education opportunities, employment opportunities in adulthood, and parental or neighbourhood role models.

Education and access to modern sector jobs are therefore among most plausible conditions for escaping poverty. Primary level education is important for sorting workers into informal sector employment, which from a policy point of view may mean that investment in primary level education is likely to place more persons – especially women – in this sector. Earnings increase with the level of education, which in turn helps to reduce poverty by increasing the productivity of the poor. Extreme poverty appears to be much more common in households that are based on informal activity – most of the working poor in Kenya are disproportionately in the informal sector (Pollin et al., 2007; Odhiambo and Manda, 2003). The proportion of working women is relatively higher in the informal sector compared with the formal sector. One conclusion of this paper is that there is need for policies to upgrade women's education and skills so that they can equally compete with men in the labour market. This would ultimately reduce their levels of unemployment and increase their productivity and earnings. Any approach to mitigating urban poverty should strongly endeavour to improve worker's productivity and earnings, particularly in the informal sector where most women and the urban poor are engaged.

The employment situation in Kenya clearly presents a daunting challenge for the Kenyan government. On the one hand, productive jobs are basically limited to wage employment in the formal sector. Returns to education are also quite high, and especially so for tertiary level education. Moreover, the changing structure of the Kenyan labour market has been marked by shifts in labour demand in favour of skilled and highly educated labour (Manda and Sen, 2004; Manda, 1997). Besides, the Kenyan economy has become increasingly capital intensive despite the abundance of labour. The private sector, which is targeted as the main conduit for employment and growth, is likely to employ high-skilled labour. This largely benefits men since they are relatively more skilled and educated than women. On the other hand, how to achieve this target remains to be seen, given a high labour force growth that outstrips employment growth and the inability of the economy to generate modern wage employment.

On the whole, these results suggest that education is very important for women (as it is for men) in enhancing their access to modern wage employment (in the private sector mainly). However, women are still more likely to attain lower levels of human capital (education, experience and skills) than men, which may put them at a greater risk of unemployment and movement into informal sector employment. Low levels of education and skills among women are mirrored in their higher unemployment rates, their underrepresentation in formal employment and their lower earnings in comparison with those of males. Wambugu's 2003 study on the importance of worker's education (among other factors) on an individual's type of employment (both urban and rural labour markets together) corroborates this argument in which women's lower human capital levels restrict them to agriculture and unpaid family work while men's higher levels of education keep them in modern wage employment. In another study Mariara (2003) also finds that there are marked differences in the process, generating the gender wage gaps in the private and public sectors of the Kenyan labour market where preferential treatment towards men is pronounced in all sectors owing to expected lower productivity of women of childbearing age.

A standard labour market is described by both the supply and the demand sides. This study focuses on the supply factors (i.e., on the characteristics of persons supplying labour to the market). A thorough understanding of the demand would be necessary to complement such a study, but relevant data on the demand side are lacking. Future DETERMINANTS OF EMPLOYMENT IN THE FORMAL AND INFORMAL SECTORS OF THE URBAN AREAS OF KENYA 21

research and survey data collection methodologies should incorporate the demand-side information.

Despite the constraints, this study identifies research, education and employment policy gaps by asking: What specific skills or qualities do employers look for when recruiting new employees? Are the recruitment practices gender balanced? Which training and skills are sought for what sectors? Is the current education system demand or supply driven, and does it equip graduates with adequate skills to become self-employed? Do the current policy environment and infrastructure encourage self-employment? What are the real constraints faced by women in finding reasonable work given their remarkably high unemployment rates? Answers to these questions have broad policy implications towards an achievement of gender balance in education, the labour market and poverty eradication. The study not only calls for further investigation into the factors influencing the demand for labour, especially in the private and informal sectors, but also asks whether there is indeed a bias towards male preference in private sector employment and whether the bias is justified on human capital attributes and not on labour market discrimination against women, especially among married women.

Finally, these policy gaps should be seen in light of the fact that women may be more likely to attain lower levels of human capital than men because of discrimination within households where families deliberately choose to invest in boys and not in girls education. This may put the girls at a greater risk of unemployment and informal sector employment, while their marital status may trigger the risk of future unemployment, inactivity or temporary interruptions from work because of reproductive responsibilities and care work. It could also make them more susceptible to sex discrimination in the labour market as employers try to safeguard production costs such as mandatory and paid maternity leave and anticipated interruption from work. Further research is thus needed to identify the actual causes.

Notes

- In the Kenyan context the "formal sector" (referred to as the modern sector) is defined to include the entire public sector and private sector enterprises and institutions that are formal in terms of registration, taxation and official recording (incorporated enterprises). The public sector covers all activities and establishments of the central government, its statutory corporations (wholly owned corporations or parastatals) and registered companies in which the government is a majority shareholder, and all local government authorities. Public sector activities are entirely in the modern economy. The private sector consists of companies and businesses in the modern sector in which the government does not own majority shares, the informal sector, cooperatives, non-profit making organizations, private households employing domestic servants, and small-scale/subsistence farming and pastoral activities. See Republic of Kenya, (1998, 2003).
- The origin of these classifications comes from literature on dual labour market and labour market segmentation models. See Ricardo (1815), Lewis (1954), Rans and Fei (1961), and Doeringer and Poire (1971).
- 3. The "informal sector" concept (now referred to as the "informal economy") was first introduced by the International Labour Organization (ILO) in the early 1970s when the term was used to describe specific activities taking place in urban areas of developing countries. The concern at that time was with the working poor who were not recognized, registered or protected by the working authority (ILO, 1972). See also Menke (1998) for a succinct discussion of the evolution of this concept. In the Kenyan context, the informal sector (locally known as the *Jua-Kali*, a Kiswahili term meaning "hot sun" to indicate that many workers operate without fixed premises) covers all small-scale activities that are normally semi-organized and unregulated and use low and simple technology. The sector largely comprises self-employed persons or employers of a few workers. It also includes unpaid family workers. Small-scale agriculture and pastoral activities are farm-related economic activities that are mainly located in the rural areas. Owing to their non-registration nature, they are not classified as belonging to either the modern sector or the informal sector (Republic of Kenya, 2003).
- 4. It is worth mentioning that non-agricultural household enterprises (NHEs) are a fundamental pillar of the Kenyan economy. According to Pollin et al. (2007) about 2.1 million NHEs existed in Kenya in 2006 of which 90% were informal enterprises and the rest formal. Out of 13.5 million Kenyans in the labour force (in 2006), 12.1 million were employed. About 43% or 5.2 million of the total employed persons work in NHEs 4.1 million (79%) in informal enterprises and 1.1 million (21%) in formal enterprises. The largest category of employment in the NHE sector is unpaid family members, comprising about 37% of total employment followed by own-account workers (people working alone

for themselves) comprising about 34% of total employment. Thus, own-account workers and unpaid family members together account for more than 70% of the entire employment in NHEs. The domestic market within Kenya is the most important source of sales from NHEs, with about 90% of them selling directly to consumers. Hardly do household enterprises sell their commodities to the export market. The overwhelming majority of NHEs – 82% of the total number of firms – derive their earnings by providing services, mainly in retail trade. Service providing enterprises comprise about 63% of all NHEs.

- 5. See also Magnac(1991) and Pradhan and van Soest (2005).
- 6. "The term can be defined as a set of policy changes or reforms which combine short-run stabilization measures and long-term adjustment measures" (van der Hoeven and van der Geest, 1999: 9). "SAP mainly entail changes in macroeconomic policies to make the economy adaptable to changing economic realities and basically more market oriented" (Ikiara and Ndung'u in van der Hoeven and van der Geest, 1999: 73). Implementation of SAPs in Kenya required restructuring of numerous public services and social support systems with the intention of reducing the government budget deficit and correcting macroeconomic imbalances for medium- and long-term economic recovery and growth. The process has also involved cost-sharing of basic social services such as health and education; retrenchments in the public sector; privatization and sale of non-strategic public enterprises; removal of price controls and subsidies; and trade liberalization.
- 7. It should be emphasized here that the linkage and dissimilarity between the formal and informal sectors is no longer as simple as it was in the 1970s and 1980s when the formal sector was viewed as offering "real" jobs with fringe benefits, job security and superior prospects (King, 1996; Agesa and Agesa, 1999). Today, the sector (mostly the private sector) also uses informal labour arrangements such as casual work arrangements an uncertain macroeconomic environment can discourage investment both domestically and externally with an adverse affect on output and employment growth. Formal enterprises suffer, and consequently are enticed to perform informal business activities alongside their official activities. The link between officially registered enterprises and various informal businesses or practices has become a common trend; informal activities in many developing countries are associated with a lack of opportunity in the formal economy and thus symbolize a coping strategy for survival.
- 8. According to Husmanns (2004: 1), the main gaps in the informal sector definition are that: 1) persons engaged in very small-scale or casual self-employment activities may not report in statistical surveys that they are self-employed, or employed at all, although their activity falls within the enterprise-based definition; 2)informal sector statistics may be affected by errors in classifying certain groups of employed persons by status of employment such as subcontractors, free-lancers or other workers whose activity is at the borderline between self-employment and wage employment; and 3) an enterprise-based definition of the informal sector is unable to capture all aspects of "informalization" of employment, which has led to a rise in various forms of informal (or non-standard, irregular, precarious, etc.) employment, in parallel to the growth of the informal sector.
- 9. Overall, Kenya's economic growth record has been sporadic and can be split into three broad phases: the growth phase (1964–1972); the shocks phase (1973–1984), and the adjustment phase (1985–2002), characterized by inconsistent donor flows and economic stagnation (particularly from 1990 to 2002) leading to an economic recovery from 2003.

- 10. Economic performance during the first half of the 1980s was frail with an average real GDP growth rate of 3.4% for the period 1980–1984. Although the economic picture in the second half of that decade was relatively better, it started declining persistently from the early 1990s. The period 1991–1993 was Kenya's worst economic performance since independence marked by a standstill in GDP growth, contraction in agricultural production at a yearly rate of 3.9%, inflation that reached a record 100% in August 1993, and a government budget deficit of more than 10% of GDP. As a result of these combined problems, bilateral and multilateral donors suspended programme aid to Kenya in 1991 and again in 1997 as a prerequisite for more reforms. There were some further shocks both internally and externally: 1992-1993 drought, deteriorating terms of trade; increases in oil prices because of the Gulf War; declining foreign investments; bad rains (El Nino) in 1997 (significantly affecting agricultural production); and two major bombings (in 1998 in Nairobi and in 2002 in Mombasa) with a heavy impact on the tourism industry. By 2000, the economic situation seemed hopeless, recording a negative real GDP growth rate. In 1993, the Government began a major agenda of economic reorganization and liberalization with the backing of the World Bank and the International Monetary Fund (IMF) - see footnote 7. Consequently, during the period 1994–1996, real GDP growth rate averaged just above 4% annually. In 1997, however, the economy regressed into sluggish growth caused by adverse weather conditions and tapering economic activity preceding general elections in December 1997. From 2004 the economy showed remarkable recovery until the post-election violence experienced after the December 2007 elections threatened to overturn this progress. Average real GDP growth rate for the period 2004 to 2006 was 5.6%. Even so, no significant employment gain was witnessed in the latest era of economic recovery. In sum, several decades of declining economic performance, combined with rapid population growth have translated over time into diminished per capita income, increased poverty and worsening unemployment.
- For instance, the incidence of urban poverty increased from 15.3% in 1976 to 29.3% in 1992, to 29% in 1994 and to 51.5% in 2000. Rural poverty increased from 25% in 1974 to 48.8% in 1981, to 46.3% in 199, to 46.8% in 1994, to 52.9% in 1997 and to 60% in 2000.
- 12. It is widely acknowledged that incomes in the informal sector are low and that they are in most cases inadequate. In the Kenyan context, Mwabu et al. (2004) show that employment in the agricultural and informal sectors is associated with a higher than average probability of being poor. Zepeda (2007) finds that even after taking into account education, sex and other factors, paid employees in the modern sector earn significantly higher wages than those in the informal sector and farm or pastoral activities. Wages in the modern public sector are highest, followed by those in the modern private sector.
- 13. Kenya was among the first African countries to adopt SAPs, back in 1980, after a prolonged economic decline, falling world commodity prices, fiscal and monetary instability, and rising levels of unemployment. See Ikiara and Ndung'u (1999) for a summary of Kenya's first 15 years of experience with structural adjustment.
- 14. Over the last three decades, Kenya has had an impressive record in expanding access to education by establishing a comprehensive network of schools all over the country. Prior to independence (in 1963), the primary school gross enrolment ratio was 47% in 1960 and had nearly doubled to 90% by 1980. The expansion of primary school enrolment was partly fuelled by free primary education initiated in 1974. Even more drastic was the expansion of secondary school enrolment rate from a mere 2% in 1960 to 19% in 1980

(more than nine-fold increase) and to about 30% in 1990, with the increase in secondary school enrolment due to a large number of schools that were built through self-help initiatives in response to high demand for education (Manda et al., 2006; Manda, 1997). One important result was that the share of illiterate females and males in the population declined over time from 44.0% in 1970 to 11.7% in 1999 for males and from 74.1% in 1970 to 25.2% in 1999 for females.

- 15. The rapid expansion of education has been accompanied by worsening unemployment problem ever since the late 1960s. Moreover, since 1989 the impressive performance during the first two decades after independence has been reversed. Primary school gross enrolment rates fell from 106% in 1989 to 101.4% in 1999 as secondary school enrolment rates declined from 26% in 1990 to 19% in 1999. Factors contributing to this decline include the high cost of education, crippling poverty, socio-cultural values, and early marriages and pregnancies. The introduction of cost-sharing arrangements in 1989 had an adverse impact on the access to education for children from poor families, especially for girls.
- 16. The basic underlying assumptions theory refer to utility maximization subject to budget constraints. The theory assumes that economic agents make informed and rational decisions on the basis of complete certainty about prices and wages and that individuals face own budgetary constraints independent of what others do. Hence, it is an application of consumer behaviour theory. Individuals are assumed to allocate time to market work and non-marketable activities (leisure). Utility is maximized by choosing a combination of goods and leisure hours subject to time, price and income constraints.
- 17. The reservation wage is the amount of money that a person would have to be given to be induced to work for the first one hour or the minimum wage at which a person is willing to enter employment. The market wage is the present value of wages offered in the market and the present value of future earnings' losses caused by non-accumulation and depreciation of human capital (Even, 1987). Therefore, it is highly dependent on level of education, accumulated work experience and length of career breaks. When the market wage is less than the reservation wage, hours of work will be zero, since the utility loss from giving up even one hour of leisure to participate in the labour force would be greater than utility gained from the income earned from market work.
- 18. For instance, the economic background of a husband or partner is expected to play an important role, especially when children are present. A conjecture of Becker's (1965) theory of household time allocation is that an increase in husband's income may prompt women to consume additional non-market time, thus implying an inverse relationship between husband's income and women's employment decisions whereby the higher the husband's income, the lower the financial pressure of the family and the lower the propensity of the wife to enter employment.
- 19. For instance, married women's labour supply decisions are typically made in the context of decisions taken by other members of the household or family. To overcome these objections, various extensions of the individual labour supply model have been made, such as game theory models, individual utility models, bargaining models and new household economics model.
- 20. A multinomial logit model is a direct extension of a binary logit model to a dependant

variable with several unordered categories. A decision to work in formal or informal sectors is not sequential/ordered but depends on the sector in which one finds a job. For instance, some people choose to join the informal sector awaiting modern wage employment job, others leave the modern sector to become self-employed or leave the public sector to join the private sector and vice versa. This choice does not assume any order justifying the use of a MNL model.

- 21. In MNL models, there is a tendency to separate as many states as is feasible (non-pooling). Cramer and Ridder (1991: 268) warn that unless the aim of the analysis calls for non-pooling, "we should pursue parsimony and pool separate states whenever possible. It is an empirical issue whether a subset of states can be treated as a single state or whether its members show significant differences, an issue subject to statistical test. The solution is to carry out a likelihood ratio test". In this study, analysis is done without pooling public and private wage employment as the two are mutually exclusive. There is no reason to pool the informal sector with modern wage employment.
- 22. See for example, Greene (2003: 667).
- 23. Age-squared is included as well to allow for non-linear relationship between age and the probability of employment.
- 24. Nevertheless, women may still be less advantaged than men in this sector partly and mainly because of the reproductive burden. For instance, age and number of younger children in a household may hinder some women in low-income households from participating in informal sector employment as wage employees or own-account workers even when they have minimal financial ability to do so. Lack of capital to start own business is yet another constraint regardless of the presence of young children.
- 25. Manda (1997) found the relationship negative but statistically insignificant, and argues that there are perhaps social mechanisms of spreading the burden of rearing children through, for instance, the presence of extended family members in a household.
- 26. Inclusion of both working and non-working persons in the analysis may lead to model specification problems since the equation is only valid in a case where the value of wages is greater than the reservation wage. Measurement problems may also be due to the large number of non-working persons whose wages are unobserved, while OLS estimates based only on the number of working persons may be biased because of a selectivity bias problem.
- 27. Labour force (economically active) refers to the number of people aged 15–64 who are either employed or unemployed.
- 28. Although migrants are predominantly males in the working age population, over the years there has been a slight change in the sex ratio of the migrants, from 121 to 119 to 114 in 1969, 1979 and 1989, respectively. A declining sex ratio suggests that the extent of female rural to urban migration has increased over time.
- 29. On average, incomes are much higher in the formal than informal sector, with more significant income disparities among the categories of workers within each sector (Pollin et al., 2007; Manda and Sen, 2004; Mwabu et al, 2003).

DETERMINANTS OF EMPLOYMENT IN THE FORMAL AND INFORMAL SECTORS OF THE URBAN AREAS OF KENYA

30. In general, although females constitute about 50.1% of the total population, on average, they account for only about 30% of the total formal sector wage employment and earn 33% less than their male counterparts (Were and Kiringai, 2004). See also Zepeda (2007) and Mariara (2003).

- 31. Based on the 1998 Integrated Labour Force Survey 1998/99 report (Republic of Kenya, 2003).
- 32. Nevertheless, women may still be less advantaged than men in this sector partly and mainly due to care work. Age and number of young children in a household may hinder some women in low-income households from participating in informal sector employment as wage employees or own-account workers even when they have a minimal financial ability to do so. A lack of capital to set up own business regardless of the presence of young children, is another major hindrance. Women in middle-income households are less constrained by care work because they can afford to hire domestic servants.
- 33. Note that if the coefficient estimate for the age-marital status interaction term is significant, it means that the *interval scale* variable *age* has a different effect on married persons relative to single persons. The coefficient estimate gives the differential effect that the *interval scale* independent variable (*age*) has on the dependent category of interest (public, private or informal sectors) versus being in the base (or contrast) category of the dependent variable (being unemployed) for the included group in the dummy variable (being married) versus the excluded group (being single). If the estimate for age-married is positive, it means that the effect of age on married persons is more positive than the effect of age on single persons (for a specific outcome category relative to being unemployed). If the coefficient estimate is negative, it would mean that the effect of age on married persons would be more negative than the effect of age on single persons (for a specific outcome category relative to being unemployed).
- 34. Results are not reported where the interaction term is significant, but are available upon request.
- 35. Women's employment is shaped by household structures and dynamics such as the size of the household, number of dependents and the household cycle; and ideological factors such as gender values and more specifically husband's attitudes to women's work. On the basis of my doctorate data collected in 2003 on employment histories and coping strategies of selected households from two different urban income clusters, I find that in the Kenyan context, the reproductive burden affects married women with children from low and middle incomes differently. The burden is seemingly less among the latter. Among women from poor urban households, early marriage and early entry into the labour market are linked to early school termination. After marriage, the reproductive duty tends to constrain them (dependent upon the number of young children below school-age) from engaging in productive work. Cultural attitudes also play a negative role. However, with a household's deteriorating economic position in the context of unstable, low-paid and precarious tenure forms of income-generating activities of their male spouses, they are pushed to work out of economic necessity. The most convenient sector to join is the informal sector as ownaccount or unpaid family workers. Others are rendered completely inactive because of young children. In contrast, women from middle-income households may voluntarily choose not to work in order to care for the children. If not, they can afford to employ childcare whenever they wish to work.

References

- Agesa, J. and R.U. Agesa. 1999. "Gender differences in the incidence of rural to urban migration, Evidence from Kenya". *Journal of Development Studies*, 35(6): 36–58.
- Blundell, R., J. Ham and C. Meghir. 1987. "Unemployment and female labour supply". *The Economic Journal*, 97: 44–64.
- Becker, G. 1965. "A theory of the allocation of time". Economic Journal, 75(299): 493-508.
- Cramer, J. S. and G. Ridder. 1991. "Pooling states in the multinomial logit model". Journal of Econometrics, Elsevier, 47(2–3): 267–72.
- Doeringer, P. and M. Poire. 1971. "International labour market and man power analysis", Lexington, Massachusetts: D.C.
- Even, W.E. 1987: "Career Interruptions Following Childbirth". Journal of Labour Economics, 5(2): 255–77.
- Gong, X. and A. van Soest. 2000. Family Structure and Female Labour Supply in Mexico. Discussion Paper No. 214. Institute for the Study of Labour (IZA), Bonn, Germany.
- Gray, M., L. Qu., D. de Vaus, C. Millward. 2002. Determinants of Australians Mothers' Employment: An Analysis of Lone and Couple Mothers. Research Paper No. 26, Australian Institute of Family Studies, Melbourne.
- Gray, M., L. Qu, F. Renda, D. de Vaus. 2003. Changes in the Labour Force Status of Lone and Couple Australian Mothers, 1983–2002. Research Paper No. 33, Australian Institute of Family Studies, Melbourne.
- Greene, William H. 2003. *Econometric Analysis*, Fifth Edition. Upper Saddle River, NJ: Prentice Hall.
- Heckman, J. 1979. "Sample selection bias as a specification error". Econometrica, 47: 153-61.
- Hunter, B.H. and M.C. Gray. 1999. "Further investigations into indigenous labour supply: what discourages discouraged workers?" Working Paper No. 2. Centre for Aboriginal Economic Policy Research (CAEPR), The Australian National University: Canberra.
- Hussmanns, R. 2004. "Measuring the informal economy: from employment in the informal sector to informal employment". Working Paper No. 53, Policy Integration Department, Bureau of Statistics, ILO, Geneva.
- Ikiara, G.K. and S.N. Ndung'u. 1999. "Employment and labour markets during adjustment Kenya". In R. Van der Hoeven and W. Van der Geest, eds., Adjustment Employment and Missing Institutions in Africa, The Experience in Eastern and Southern Africa. Geneva: International Labour Organization.
- ILO. 1972. Employment, Incomes and Equality: A Strategy for Increasing Productive Employment in Kenya. Geneva: International Labour Organization.
- Kaufman, B. E. 1994. The Economics of Labour Markets. Fort Worth: Dryden Press.
- Killingsworth, M. 1983. Labour Supply. Cambridge: Cambridge University Press
- King, K. 1996. "Micro enterprise, macroeconomic environment: Revisiting Kenya's informal (jua kali) sector against the background of the formal globalizing economy". *International*

Journal of Educational Development, 16(4): 417–26.

- Kolev, A. 1998. Labour Supply in the Informal Economy in Russia during Transition. Discussion Paper No. 2024. Centre for Economic Policy Research, London.
- Lanot, G. and C. Muller. 1997. "Dualistic sector choice and female labour supply: Evidence from formal and informal sectors in Cameroon". Working Paper Series/97-9. Centre for the Study of African Economies, University of Oxford.
- Lewis, W.A. 1954. "Economic Development with Unlimited Supplies of Labour", *The Manchester School of Economics and Social Studies*, 22(2): 139–91.
- Magnac, T. 1991. "Segmented or competitive labour markets". Econometrica, 59(1): 165-87.
- Makonnen, N. 1993. *Labour Supply and the Distribution of the Economic Well-being, A Case Study of Lesotho.* Ph.D Thesis, University of Gothenburg, Sweden.
- Manda, D.M. 1997. Labour Supply, Returns to Education, and the Effect of Firm Size on Wages: The case of Kenya. Ph.D Thesis, University of Gothenburg, Sweden.
- Manda, D.K. and K. Sen. 2004. "The labour market effects of globalization in Kenya"
- Journal of International Development, 16: 29–43.
- Manda, D.K., G. Mwabu, S.M. Kimenyi. 2006. "Human capital externality and returns to education in Kenya." *Eastern Economic Journal*, 32: 493–514.
- Mariara, J.K. 2003. Wage Determination and Gender Wage Gap in Kenya, Any Evidence of Gender Discrimination? Research Paper No. 132. African Economic Research Consortium, Nairobi.
- Mazumdar, D. 1989. *Microeconomic Issues of Labour Markets in Developing Countries: Analysis and Policy Implications*. Washington, D.C.: The World Bank.
- Menke, J. 1998. *Restructuring, Urban Employment and Poverty: The Case of Surinam*. Pamaribo: SWI Press.
- Milne, W. J., G.M. Barber, G. Ongile. 1990. "Determinants of Urban Labour Force Participation in Kenya", *Eastern Africa Economic Review*, 6(2): 83–93.
- Milne, W. J., M. Neitzer. 1994. "Kenya". In, Horton, S, R. Kanbur, D. Mazumdar, *Labour Market* in the Era of Structural Adjustment, Vol. 2. Washington, D.C.: The World Bank.
- Mwabu, G.M., S.M. Kimenyi, P.K. Kimalu, N. Nafula, D.K. Manda. 2003. "Predicting Household Poverty; a Methodological Note", *African Development Review*, 15(1):77–85.
- Mwabu, G., M. Oiro and D.K. Manda. 2004. *Poverty and Employment in Kenya*. Discussion Paper No. 33. Kenya Institute of Public Policy Research and Analysis (KIPPRA), Nairobi.
- Myint, H. 1985. "Organizational dualism and economic development". *Asian Development Review*, 3(1): 24–42.
- Nwanganga, S. 1980. "Women in the urban labour markets of Africa: The case of Tanzania". Staff Working Paper No. 380. The World Bank, Washington, D.C.
- Odhiambo, W. and D.K. Manda. 2003. Urban poverty and labour force participation in Kenya".
- Paper presented at the World Bank Urban Research Symposium, Washington, D.C., 15–17 December.
- Pollin, R., M. Githinji and J. Heintz. 2007. An Employment Targeted Economic Program for Kenya. Political Economy Research Institute, Amherst, USA.
- Pradhan, M., A. van Soest. 1995. "Formal and informal sector employment in urban areas of Bolivia". *Labour Economics*, 2:275–97.
- Ranis, G. and J. Fei. 1961. "A theory of economic development". American Economic Review, 51(4): 533–65
- Republic of Kenya. 1985. Sessional Paper No. 2 of 1985 on Unemployment. Nairobi: Government Printer.
- Republic of Kenya. 1986. Sessional Paper No. 1 of 1986 on Economic Management for Renewed Growth. Nairobi: Government Printer.
- Republic of Kenya. 1988. Urban Labour Force Survey, 1986. Nairobi: Government Printer.

Republic of Kenya. 1998. Economic Survey. Nairobi: Government Printer.

- Republic of Kenya. 2003. Report of 1998/99 Labour Force Survey Report. Nairobi: Government Printer.
- Republic of Kenya. Various Issues. *Economic Survey*. Central Bureau of Statistics, Ministry of Planning and National Development.
- Ricardo, D. 1815. *Works and Correspondence of David Ricardo*, Vol. 1–4. Cambridge: Cambridge University Press.
- Shoshana, N., and G. Shoshana 2008. "The Extra Burden of Moslem Wives: Clues from Israeli Women's Labour Supply". *Economic Development and Cultural Change*, 46(3): 491–517.
- Van den Brink, H.M. 1994. Female Labour Supply, Child Care and Marital Conflict. Amsterdam: Amsterdam University Press.
- Van der Hoeven, R. and W. Van der Geest, eds. 1999. Adjustment Employment and Missing Institutions in Africa: The Experience in Eastern and Southern Africa. Geneva: International Labour Organization.
- Wambugu, A. 2003. *Essays on Earnings and Human Capital in Kenya. Department of Economics*, Doctorate Dissertation, Guteborg, Sweden.
- Were, M., and J. Kiringai. 2004. "Gender mainstreaming in macroeconomic policies and poverty reduction strategy in Kenya". African Women's Development & Communication Network, Nairobi. At http://www.femnet.or.ke/documents/gender_mainstreaming.pdf
- Zepeda, E. 2007. "Addressing the employment-poverty nexus in Kenya: Comparing cash-transfer and job-creation programmes". Working Paper No. 40. International Poverty Centre, Brasilia.

Appendix A: Variable descriptions and characteristics

Variable description

Table A1: Explanatory variables

Age

Age-squared (agesq)

Head=whether one is a household head or not, dummy variable in which 1=household head; 0=not household head

Household size=total number of household members (hsize)

Sex= male or female, dummy variable in which 1=male; 0=female

Married =whether married or not, dummy variable in which 1=married; 0=not married

AgeMarried, Interaction dummy for age * Married

Education (highest level completed)

Primary, dummy variable in which 1=has primary level education; 0=other level or none

Secondary, dummy variable in which 1=has secondary level education; 0=other level or none

University, dummy variable in which 1=has university level education; 0=other level or none

None/nursery (*omitted category*), dummy variable in which 1=has no schooling; 0=has primary, secondary and university

Presence of children and female relatives in a household

Presence of children 0-6 years in household, dummy variable in which 1=yes; 0=no (Ch0_6)

Presence of children 7–17 years, dummy variable in which 1=yes; 0=no (Ch7_17)

Presence of female relatives in a household, dummy variable in which 1=yes; 0=no (Relatives)

Ownership of a dwelling unit= dummy variable in which 1=owns a house or 0 does not (Tenure)

Log monthly wage earnings Salary,

Table A2: Tota	il samp	le										
Variable		Public			Private			Informal		Ū	nemploye	q
	Obs	Mean	Std. Dev.	Obs	Mean	Std. Dev.	Obs	Mean	Std. Dev.	Obs	Mean	Std. Dev.
Age	537	36.42	7.63	724	33.30	9.31	1017	33.18	10.19	096	28.69	10.67
Agesq	537	1384.64	582.11	724	1195.50	670.87	1017	1204.96	753.39	096	936.64	767.32
Head	537	0.78	0.41	724	0.77	0.42	1017	0.61	0.49	096	0.18	0.38
Sex	537	0.67	0.47	724	0.74	0.44	1017	0.53	0.50	960	0.26	0.44
Hsize	537	3.87	2.21	724	3.64	2.41	1017	4.19	2.42	960	4.71	2.64
Married	537	0.84	0.37	724	0.70	0.46	1017	0.69	0.46	960	0.64	0.48
AgeMarried	537	31.23	15.25	724	25.29	17.89	1017	24.41	18.15	960	19.57	17.04
None	537	0.01	0.11	724	0.04	0.19	1017	0.09	0.29	960	0.11	0.31
Primary	537	0.15	0.35	724	0.32	0.47	1017	0.47	0.50	960	0.45	0.50
Secondary	537	0.76	0.43	724	0.59	0.49	1017	0.41	0.49	960	0.43	0.50
University	537	0.08	0.27	724	0.05	0.21	1017	0.02	0.14	960	0.02	0.12
Ch0_6	537	0.40	0.49	724	0.39	0.49	1017	0.43	0.50	960	0.47	0.50
Ch7_17	537	0.44	0.50	724	0.34	0.47	1017	0.46	0.50	960	0.42	0.49
Relatives	537	0.18	0.39	724	0.14	0.35	1017	0.15	0.36	960	0.22	0.42
Tenure	537	0.14	0.35	724	0.12	0.32	1017	0.16	0.37	960	0.17	0.37
Salary	511	8.84	0.59	622	8.56	0.88	270	8.05	0.75	0		
Proportion total	16.6			22.4			31.4			29.7		

Sample characteristics

32

RESEARCH PAPER 194

Table A3: Male	∋s – Tot	tal sample										
Variable		Public			Private			Informal			Jnemployed	_
	Obs	Mean	Std. Dev.	Obs	Mean	Std. Dev.	Obs	Mean	Std. Dev.	Obs	Mean	Std. Dev.
Age	360	37.67	7.83	539	34.45	9.04	542	34.24	10.54	250	30.23	12.29
Agesq	360	1480.08	612.25	539	1268.18	669.52	542	1283.03	794.42	250	1064.25	898.26
Head	360	0.96	0.21	539	06.0	0.30	542	0.83	0.38	250	0.44	0.50
Hsize	360	3.71	2.26	539	3.53	2.39	542	3.98	2.54	250	4.69	2.68
Married	360	0.92	0.27	539	0.81	0.40	542	0.75	0.44	250	0.41	0.49
AgeMarried	360	35.36	12.71	539	29.17	16.25	542	27.80	18.25	250	16.66	21.26
None	360	0.01	0.12	539	0.04	0.19	542	0.06	0.24	250	0.08	0.27
Primary	360	0.18	0.38	539	0.31	0.46	542	0.48	0.50	250	0.42	0.49
Secondary	360	0.70	0.46	539	0.61	0.49	542	0.42	0.49	250	0.48	0.50
University	360	0.11	0.31	539	0.05	0.21	542	0.03	0.17	250	0.02	0.15
Ch0_6	360	0.40	0.49	539	0.39	0.49	542	0.41	0.49	250	0.32	0.47
Ch7_17	360	0.14	0.35	539	0.11	0.31	542	0.11	0.31	250	0.18	0.38
Relatives	360	0.11	0.32	539	0.11	0.31	542	0.16	0.36	250	0.19	0.39
Salary	342	8.92	0.61	472	8.67	0.81	196	8.22	0.65	0		
Proportion total	21.3			31.9			32.1			14.8		

DETERMINANTS OF EMPLOYMENT IN THE FORMAL AND INFORMAL SECTORS OF THE URBAN AREAS OF KENYA

Table A4: Fema	ales – t	total samp	ple									
Variable		Public			Private			Informal			nemploye	T
	Obs	Mean	Std. Dev.	Obs	Mean	Std. Dev.	Obs	Mean	Std. Dev.	Obs	Mean	Std. Dev.
Age	177	33.88	6.54	185	29.96	9.32	475	31.99	9.64	710	28.14	9.99
Agesq	177	1190.51	459.16	185	983.76	630.29	475	1115.88	693.83	710	891.71	710.81
Head	177	0.43	0.50	185	0.41	0.49	475	0.36	0.48	710	0.09	0.28
Hsize	177	4.19	2.07	185	3.99	2.45	475	4.44	2.25	710	4.71	2.63
Married	177	0.68	0.47	185	0.41	0.49	475	0.63	0.48	710	0.71	0.45
AgeMarried	177	22.81	16.50	185	13.98	17.71	475	20.54	17.25	710	20.60	15.16
None	177	0.01	0.11	185	0.04	0.19	475	0.13	0.33	710	0.11	0.32
Primary	177	0.09	0.29	185	0.36	0.48	475	0.46	0.50	710	0.46	0.50
Secondary	177	0.87	0.34	185	0.55	0.50	475	0.40	0.49	710	0.41	0.49
University	177	0.03	0.17	185	0.05	0.22	475	0.01	0.09	710	0.01	0.11
Ch0_6	177	0.41	0.49	185	0.37	0.48	475	0.47	0.50	710	0.52	0.50
Ch7_17	177	0.56	0.50	185	0.43	0.50	475	0.56	0.50	710	0.42	0.49
Relatives	177	0.25	0.44	185	0.25	0.43	475	0.20	0.40	710	0.24	0.43
Tenure	177	0.19	0.39	185	0.14	0.35	475	0.16	0.37	710	0.16	0.37
Salary	169	8.67	0.52	150	8.23	0.99	74	7.60	0.81	0		
Proportion total	11.4			12			30.7			45.9		

Table A5: Tota	l sample	e 15–34 a	ge cohort									
Variable		Public			Private			Informal		D	nemployed	T
Ι	obs	Mean	Std. Dev.	Obs	Mean	Std. Dev.	obs	Mean	Std. Dev.	Obs	Mean S	std. Dev.
Age	230	29.37	3.25	432	26.92	4.47	595	26.17	4.71	728	23.56	4.53
Agesq	230	873.08	183.33	432	744.47	234.65	595	707.27	240.22	728	575.43	223.62
Head	230	0.67	0.47	432	0.69	0.46	595	0.49	0.50	728	0.09	0.29
Sex	230	0.58	0.49	432	0.69	0.46	595	0.50	0.50	728	0.23	0.42
Hsize	230	3.60	1.99	432	3.46	2.19	595	3.97	2.32	728	4.67	2.62
Married	230	0.77	0.42	432	0.57	0.50	595	0.60	0.49	728	0.58	0.49
AgeMarried	230	22.87	12.94	432	16.40	14.45	595	16.86	13.98	728	14.42	12.70
None	230	0.01	0.11	432	0.02	0.14	595	0.03	0.18	728	0.07	0.25
Primary	230	0.07	0.26	432	0.31	0.46	595	0.51	0.50	728	0.47	0.50
Secondary	230	0.83	0.38	432	0.64	0.48	595	0.45	0.50	728	0.46	0.50
University	230	0.08	0.28	432	0.03	0.18	595	0.00	0.04	728	0.01	0.10
Ch0_6	230	0.49	0.50	432	0.41	0.49	595	0.50	0.50	728	0.52	0.50
Ch7_17	230	0.26	0.44	432	0.23	0.42	595	0.36	0.48	728	0.38	0.49
Relatives	230	0.20	0.40	432	0.18	0.39	595	0.16	0.37	728	0.23	0.42
Tenure	230	0.10	0.29	432	0.09	0.29	595	0.10	0.30	728	0.13	0.34
Salary	220	8.74	0.52	369	8.40	0.83	177	7.89	0.72	0		
Proportion total	11.6			21.8				30		36.7		

DETERMINANTS OF EMPLOYMENT IN THE FORMAL AND INFORMAL SECTORS OF THE URBAN AREAS OF KENYA

Table A6:Total	sample	e 35–64 ag	je cohort									
Variable		Public			Private			Informal			Inemployed	
	Obs	Mean	Std. Dev.	Obs	Mean	Std. Dev.	Obs	Mean	Std. Dev.	Obs	Mean 3	Std. Dev.
Age	307	41.70	5.37	292	42.74	5.99	422	43.07	7.20	232	44.78	8.05
Agesq	307	1767.89	473.65	292	1862.78	536.54	422	1906.69	666.87	232	2070.11	765.14
Head	307	0.87	0.34	292	0.89	0.31	422	0.78	0.41	232	0.46	0.50
Sex	307	0.74	0.44	292	0.83	0.37	422	0.59	0.49	232	0.34	0.47
Hsize	307	4.06	2.35	292	3.91	2.70	422	4.50	2.52	232	4.83	2.70
Married	307	06.0	0.31	292	06.0	0.30	422	0.81	0.39	232	0.81	0.39
AgeMarried	307	37.48	13.80	292	38.43	13.99	422	35.05	18.02	232	35.75	18.73
None	307	0.01	0.11	292	0.06	0.24	422	0.18	0.38	232	0.23	0.42
Primary	307	0.20	0.40	292	0.34	0.48	422	0.42	0.49	232	0.40	0.49
Secondary	307	0.70	0.46	292	0.52	0.50	422	0.36	0.48	232	0.34	0.47
University	307	0.08	0.27	292	0.07	0.26	422	0.05	0.21	232	0.03	0.17
Ch0_6	307	0.33	0.47	292	0.35	0.48	422	0.34	0.47	232	0.31	0.46
Ch7_17	307	0.57	0.50	292	0.49	0.50	422	0.61	0.49	232	0.53	0.50
Relatives	307	0.17	0.38	292	0.09	0.28	422	0.14	0.35	232	0.21	0.41
Tenure	307	0.17	0.38	292	0.15	0.36	422	0.25	0.43	232	0.28	0.45
Salary	291	8.91	0.64	253	8.80	0.90	93	8.34	0.72	0		
Proportion total	24.5			23.3			33.7			18.5		

Table A7: Male	s 15–34	age cohort										
Variable		Public			Private			Informal		IJ	nemployed	T
	Obs	Mean	Std. Dev.	Obs	Mean	Std. Dev.	Obs	Mean	Std. Dev.	Obs	Mean S	std. Dev.
Age	133	29.65	3.11	296	27.65	4.05	295	26.39	4.71	171	22.94	4.67
Agesq	133	888.99	177.00	296	780.98	218.77	295	718.72	240.16	171	547.71	230.17
Head	133	0.90	0.30	296	0.84	0.37	295	0.72	0.45	171	0.22	0.41
Hsize	133	3.15	1.84	296	3.23	2.00	295	3.59	2.43	171	4.74	2.61
Married	133	0.80	0.40	296	0.69	0.46	295	0.58	0.49	171	0.19	0.39
AgeMarried	133	24.05	12.42	296	19.81	13.71	295	16.68	14.35	171	5.08	10.85
None	133	0.02	0.12	296	0.02	0.14	295	0.02	0.14	171	0.03	0.17
Primary	133	0.09	0.29	296	0.29	0.45	295	0.53	0.50	171	0.45	0.50
Secondary	133	0.77	0.42	296	0.66	0.48	295	0.45	0.50	171	0.50	0.50
University	133	0.12	0.33	296	0.03	0.18	295	00.0	0.00	171	0.02	0.13
Ch0_6	133	0.43	0.50	296	0.41	0.49	295	0.41	0.49	171	0.32	0.47
Ch7_17	133	0.13	0.34	296	0.17	0.37	295	0.22	0.42	171	0.37	0.49
Relatives	133	0.11	0.31	296	0.13	0.34	295	0.11	0.32	171	0.18	0.38
Tenure	133	0.05	0.21	296	0.09	0.28	295	0.11	0.31	171	0.16	0.37
Salary	127	8.84	0.50	259	8.52	0.73	120	8.09	0.61	0		
Proportion total	14.9			33.1			33			19.1		

DETERMINANTS OF EMPLOYMENT IN THE FORMAL AND INFORMAL SECTORS OF THE URBAN AREAS OF KENYA 37

Table A8: Male	es 35–6	4 age coho	ort									
Variable		Public			Private			Informal		'n	nemployed	
	Obs	Mean	Std. Dev.	Obs	Mean	Std. Dev.	Obs	Mean	Std. Dev.	Obs	Mean	std. Dev.
Age	227	42.37	5.63	243	42.72	6.04	247	43.60	7.48	79	46.01	8.12
Agesq	227	1826.40	501.07	243	1861.64	542.42	247	1957.01	693.54	79	2182.34	780.52
Head	227	0.99	0.11	243	0.97	0.17	247	0.96	0.20	79	0.91	0.29
Hsize	227	4.04	2.42	243	3.89	2.76	247	4.44	2.58	79	4.58	2.85
Married	227	0.99	0.09	243	0.95	0.22	247	0.94	0.24	79	06.0	0.30
AgeMarried	227	42.00	6.86	243	40.57	10.97	247	41.07	12.70	79	41.72	16.00
None	227	0.01	0.11	243	0.06	0.23	247	0.11	0.31	79	0.19	0.39
Primary	227	0.22	0.42	243	0.33	0.47	247	0.43	0.50	79	0.34	0.48
Secondary	227	0.66	0.47	243	0.55	0.50	247	0.39	0.49	79	0.43	0.50
University	227	0.10	0.30	243	0.07	0.25	247	0.07	0.25	79	0.04	0.19
Ch0_6	227	0.38	0.49	243	0.37	0.48	247	0.40	0.49	79	0.33	0.47
Ch7_17	227	0.52	0.50	243	0.47	0.50	247	0.57	0.50	79	0.51	0.50
Relatives	227	0.17	0.37	243	0.08	0.27	247	0.11	0.31	79	0.19	0.39
Tenure	227	0.15	0.36	243	0.14	0.35	247	0.22	0.41	62	0.24	0.43
Salary	215	8.97	0.66	213	8.85	0.87	76	8.41	0.66	0		
Proportion total	28.5			30.5			31.0			9.9		

Table A9: Fema	ales 15-	-34 age c	ohort									
Variable		Public			Private			Informal		Un	employed	
	Obs	Mean	Std. Dev.	Obs	Mean	Std. Dev.	Obs	Mean	Std. Dev.	Obs	Mean S	td. Dev.
Age	97	28.98	3.40	136	25.32	4.93	300	25.96	4.71	557	23.75	4.48
Agesq	97	851.27	190.43	136	665.01	248.83	300	696.01	240.14	557	583.94	221.08
Head	97	0.34	0.48	136	0.37	0.48	300	0.27	0.44	557	0.05	0.22
Hsize	97	4.23	2.01	136	3.97	2.48	300	4.35	2.15	557	4.64	2.63
Married	97	0.72	0.45	136	0.32	0.47	300	0.62	0.49	557	0.70	0.46
AgeMarried	97	21.27	13.52	136	8.99	13.23	300	17.03	13.62	557	17.29	11.83
None	97	0.01	0.10	136	0.02	0.15	300	0.05	0.21	557	0.08	0.27
Primary	97	0.05	0.22	136	0.34	0.47	300	0.50	0.50	557	0.47	0.50
Secondary	97	0.91	0.29	136	0.61	0.49	300	0.45	0.50	557	0.44	0.50
University	97	0.03	0.17	136	0.03	0.17	300	0.00	0.06	557	0.01	0.09
Ch0_6	97	0.58	0.50	136	0.42	0.50	300	09.0	0.49	557	0.58	0.49
Ch7_17	97	0.44	0.50	136	0.38	0.49	300	0.50	0.50	557	0.39	0.49
Relatives	97	0.32	0.47	136	0.29	0.46	300	0.21	0.41	557	0.24	0.43
Tenure	97	0.16	0.37	136	0.11	0.31	300	0.09	0.29	557	0.12	0.33
Salary	93	8.61	0.50	110	8.12	0.96	57	7.47	0.75	0		
Proportion Total	8.9			12.5			27.5			51.1		

hor
00
age
15–34
es
Fema
A9:
able

DETERMINANTS OF EMPLOYMENT IN THE FORMAL AND INFORMAL SECTORS OF THE URBAN AREAS OF KENYA 39

Table A10: Ferr	nales 3	5-64 age	cohort									
Variable		Public			Private			Informal		D	nemployed	
	Obs	Mean	Std. Dev.	Obs	Mean	Std. Dev.	Obs	Mean	Std. Dev.	Obs	Mean \$	Std. Dev.
Age	80	39.83	4.00	49	42.84	5.85	175	42.31	6.74	153	44.15	7.96
Agesq	80	1601.85	336.13	49	1868.47	511.70	175	1835.66	622.33	153	2012.16	753.10
Head	80	0.54	0.50	49	0.51	0.51	175	0.53	0.50	153	0.23	0.42
Hsize	80	4.14	2.16	49	4.04	2.39	175	4.59	2.42	153	4.95	2.62
Married	80	0.63	0.49	49	0.65	0.48	175	0.63	0.48	153	0.76	0.43
AgeMarried	80	24.68	19.45	49	27.84	21.04	175	26.55	20.83	153	32.66	19.33
None	80	0.01	0.11	49	0.08	0.28	175	0.27	0.44	153	0.25	0.43
Primary	80	0.14	0.35	49	0.43	0.50	175	0.41	0.49	153	0.43	0.50
Secondary	80	0.83	0.38	49	0.39	0.49	175	0.31	0.46	153	0.29	0.46
University	80	0.03	0.16	49	0.10	0.31	175	0.02	0.13	153	0.03	0.16
Ch0_6	80	0.20	0.40	49	0.24	0.43	175	0.24	0.43	153	0.30	0.46
Ch7_17	80	0.71	0.46	49	0.57	0.50	175	0.66	0.48	153	0.54	0.50
Relatives	80	0.18	0.38	49	0.12	0.33	175	0.19	0.39	153	0.22	0.41
Tenure	80	0.21	0.41	49	0.22	0.42	175	0.29	0.45	153	0.31	0.46
Salary	76	8.74	0.53	40	8.52	1.03	17	8.01	0.87	0		
Proportion Total	17.5			10.7			38.3			33.5		

Appendix B: Results

Table B1: Total sample

(Note: * p<.05; ** p<.01; *** p<.001)

Number of obs, 3238; LR chi2(39), 1503.46; Prob > chi2, 0.00; Log likelihood, -3642.53; Pseudo R2, 0.17

	Pub	lic	Priva	te	Infori	mal Un	employed
Variable	Coef.	ME	Coef.	ME	Coef.	ME	ME
Age	0.60***	0.04	0.24***	0.01	0.19***	0.00	-0.05
Age squared	-0.01***	0.00	0.00***	0.00	0.00***	0.00	0.00
Head	2.08***	0.07	2.05***	0.15	1.74***	0.14	-0.36
Household size	-0.09	-0.01	-0.04	0.00	-0.03	0.00	0.01
Sex	0.42*	-0.01	1.12***	0.15	0.41***	-0.03	-0.12
Married	0.47*	0.05	-0.14	-0.03	-0.04	-0.01	0.00
Primary	1.47***	0.11	0.84***	0.07	0.43*	-0.05	-0.13
Secondary	2.93***	0.25	1.31***	0.12	0.20	-0.20	-0.18
University	3.30***	0.51	1.52***	0.00	0.03	-0.31	-0.20
Tenure	-0.05	0.00	-0.12	-0.02	0.01	0.02	0.01
Ch0_6	-0.18	-0.01	-0.13	-0.02	-0.02	0.02	0.01
Ch7_17	0.21	0.00	0.17	-0.02	0.48***	0.09	-0.06
Relatives	0.66***	0.06	0.24	0.02	0.07	-0.04	-0.04
Constant	-15.16***		-6.46***		-4.13***		

Table B2: Total sample 15–34 age cohort

(Note: * p<.05; ** p<.01; *** p<.001)

Number of obs, 1985; LR chi2(39), 985.07; Prob > chi2, 0.00; Log likelihood, -2109.06; Pseudo R2, 0.19

	Pub	lic	Priva	ate	Infor	mal L	Inemployed
Variable	Coef.	ME	Coef.	ME	Coef.	ME	ME
Age	0.87**	0.05	-0.09	-0.03	-0.02	-0.02	-0.01
Age squared	-0.01	0.00	0.00	0.00	0.00	0.00	0.00
Head	2.09***	0.03	2.50***	0.24	1.93***	0.15	-0.42
Household size	0.00	0.00	-0.02	0.00	-0.04	-0.01	0.01
Sex	0.36	0.00	0.98***	0.13	0.49	0.02	-0.14
Married	0.32	0.02	-0.30	-0.05	-0.15	-0.01	0.04
Primary	0.38	-0.02	0.99*	0.08	1.05***	0.14	-0.20
Secondary	2.32***	0.09	1.48***	0.16	0.72*	-0.01	-0.24
University	2.64***	0.36	1.14	0.16	-2.01	-0.36	-0.15
Tenure	0.24	0.01	0.12	0.02	-0.05	-0.03	-0.01
Ch0_6	-0.23	-0.02	-0.06	-0.02	0.15	0.05	-0.01
Ch7_17	-0.16	-0.02	0.12	-0.01	0.38	0.08	-0.05
Relatives	0.50*	0.02	0.43*	0.06	0.10	-0.03	-0.05
Constant	-18.91***		-2.87		-2.37		

Table B3: Total sample 35–64 age cohort

(Note: * p<.05; ** p<.01; *** p<.001)

INUITIDEF OF ODS, 1253, ER CHI2(39), 431.03, FTOD > CHI2, 0.00, EQUIREIITIOUU, -1401.01, FSEUDU RZ, 0.13	Number of obs,	, 1253; LR chi2(39)	, 451.65; Prob >	chi2, 0.00; Log likelihood,	-1481.81; Pseudo R2, 0.13
----------------------------------------------------------------------------------------------------------	----------------	---------------------	------------------	-----------------------------	---------------------------

	Pul	blic	Priv	ate	Inform	mal	Unemployed
Variable	Coef.	ME	Coef.	ME	Coef.	ME	ME
Age	0.65**	0.08	0.56**	0.07	-0.04	-0.12	-0.04
Age squared	-0.01***	0.00	-0.01***	0.00	0.00	0.00	0.00
Head	2.09***	0.14	1.52***	0.05	1.55***	0.10	-0.29
Household size	-0.16*	-0.02	-0.07	0.00	-0.02	0.01	0.01
Sex	0.26	-0.04	1.25***	0.18	0.25	-0.07	-0.07
Married	0.94***	0.10	0.36	0.00	0.23	-0.05	-0.06
Primary	1.82***	0.30	0.64	0.00	0.02	-0.21	-0.09
Secondary	3.06***	0.46	1.12***	0.03	-0.11	-0.36	-0.13
University	3.57***	0.57	1.73***	-0.08	0.43	-0.34	-0.15
Tenure	-0.18	-0.02	-0.33	-0.05	0.05	0.06	0.01
Ch0_6	-0.31	-0.02	-0.17	0.02	-0.34	-0.04	0.04
Ch7_17	0.53	0.04	0.11	-0.06	0.58*	0.08	-0.06
Relatives	0.62*	0.14	-0.33	-0.08	-0.04	-0.04	-0.01
Constant	-17.02***		-13.72***		1.20		

Table B4: Males total sample

(Note: * p<.05; ** p<.01; *** p<.001)

Number of obs	, 1691; L	LR chi2(36),	544.76;	Prob > chi2	, 0.00; Loo	g likelihood	, -1995.40	:Pseudo R2	, 0.12
---------------	-----------	--------------	---------	-------------	-------------	--------------	------------	------------	--------

	Pub	lic	Priva	te	Infor	mal U	nemployed
Variable	Coef.	ME	Coef.	ME	Coef.	ME	ME
Age	0.51***	0.05	0.21***	0.01	0.09	-0.04	-0.02
Age squared	-0.01***	0.00	0.00***	0.00	0.00*	0.00	0.00
Head	1.90***	0.06	1.85***	0.13	1.68***	0.08	-0.27
Household size	-0.03	0.00	-0.01	0.00	0.02	0.01	0.00
Married	1.29***	0.08	0.84**	0.04	0.69**	-0.02	-0.10
Primary	1.40*	0.11	0.71	-0.02	0.72*	-0.02	-0.08
Secondary	2.27***	0.24	0.91*	0.05	0.17	-0.20	-0.09
University	3.31***	0.50	1.38*	-0.13	0.51	-0.27	-0.10
Tenure	-0.23	-0.03	-0.11	-0.03	0.11	0.05	0.00
Ch0_6	-0.24	-0.02	-0.15	-0.01	-0.05	0.02	0.01
Ch7_17	-0.26	-0.03	-0.20	-0.04	0.06	0.05	0.01
Relatives	0.38	0.07	-0.04	-0.01	-0.20	-0.06	0.00
Constant	-12.99***		-4.85***		-2.28*		

Table B5: Males 15–34 age cohort

(Note: * p<.05; ** p<.01; *** p<.001)

Number of obs, 895; LR chi2(36), 401.55; Prob > chi2, 0.00; Log likelihood, -990.74; Pseudo R2, 0.17

	Pub	lic	Priv	ate	Infor	mal Ur	nemployed
Variable	Coef.	ME	Coef.	ME	Coef.	ME	ME
Age	0.81	0.07	0.22	0.01	-0.02	-0.04	-0.04
Age squared	-0.01	0.00	0.00	0.00	0.00	0.00	0.00
Head	2.12***	0.05	2.28***	0.29	1.97***	0.06	-0.40
Household size	0.13	0.01	0.01	-0.01	0.06	0.00	-0.01
Married	1.13**	0.05	0.88**	0.09	0.66*	-0.01	-0.14
Primary	-0.29	-0.06	0.27	0.01	0.77	0.10	-0.04
Secondary	1.09	0.08	0.50	0.05	0.13	-0.05	-0.08
University	2.11	0.40	0.48	0.05	-35.01	-0.41	-0.04
Tenure	-0.33	-0.05	0.25	0.06	0.19	0.01	-0.02
Ch0_6	-0.72	-0.05	-0.33	-0.02	-0.19	0.02	0.05
Ch7_17	-0.55	-0.05	-0.02	0.04	-0.11	0.00	0.02
Relatives	0.03		0.29	0.08	-0.15	-0.05	-0.02
Constant	-16.47*		-5.01		-1.34		

Table B6: Males 35–64 age cohort

(Note: * p<.05; ** p<.01; *** p<.001)

Number of obs, 796; LR chi2(36), 146.55; Prob > chi2, 0.00; Log likelihood, 971.40; Pseudo R2, 0.07

Variable	Pub	olic	Priv	/ate	Info	rmal Uı	nemployed
	Coef.	ME	Coef.	ME	Coef.	ME	ME
Age	0.65*	0.09	0.50*	0.06	-0.07	-0.12	-0.03
Age squared	-0.01**	0.00	-0.01*	0.00	0.00	0.00	0.00
Head	1.67*	0.14	0.99	0.03	0.81	-0.03	-0.13
Household size	-0.12	-0.02	-0.03	0.01	-0.02	0.01	0.00
Married	2.42**	0.22	0.72	-0.02	0.51	-0.10	-0.10
Primary	1.98**	0.27	0.85	-0.06	0.63	-0.13	-0.08
Secondary	2.66***	0.38	1.07*	-0.01*	0.19	-0.27	-0.10
University	3.71**	0.52	1.81*	-0.17*	1.11	-0.26	-0.10
Tenure	-0.22	-0.02	-0.36	-0.06	0.04	0.07	0.01
Ch0_6	0.16	0.01	0.15	0.01	0.12	0.00	-0.01
Ch7_17	0.05	0.00	-0.23	-0.08	0.29	0.08	0.00
Relatives	0.32	0.17	-0.70	-0.13	-0.46	-0.07	0.02
Constant	-18.02**		-11.14*		1.99		

Table B7a: Females total sample without age-marital status interaction term

(Note: * p<.05; ** p<.01; *** p<.001)

Number of obs, 1547; LR chi2(36), 587.49; Prob > chi2, 0.00; Log likelihood, -1596.68; Pseudo R2, 0.16

Variable	Pul	blic	Priv	ate	Infor	mal Un	employed
	Coef.	ME	Coef.	ME	Coef.	ME	ME
Age	0.78***	0.03	0.19**	0.01	0.22***	0.03	-0.07
Age squared	-0.01***	0.00	0.00**	0.00	0.00***	0.00	0.00
Head	1.75***	0.04	1.36***	0.05	1.61***	0.26	-0.35
Household size	-0.15*	-0.01	-0.07	0.00	-0.05	-0.01	0.02
Married	0.11	0.01	-1.13***	-0.13	-0.18	0.01	0.10
Primary	1.27	0.05	1.16***	0.11	0.25	-0.02	-0.14
Secondary	3.53***	0.19	1.68***	0.14	0.18	-0.11	-0.22
University	2.77***	0.19	2.60***	0.42	-0.58	-0.29	-0.32
Tenure	0.30	0.02	-0.20	-0.02	-0.08	-0.01	0.02
Ch0_6	-0.16	-0.01	-0.07	-0.01	0.04	0.01	0.00
Ch7_17	0.45	0.01	0.33	0.01	0.63**	0.12	-0.13
Relatives	0.69*	0.03	0.11	0.00	0.14	0.01	-0.05
Constant	-18.04***		-5.28***		-4.65***		

Table B7b: Females total sample controlling for age-marital status interaction term

(Note: * p<.05; ** p<.01; *** p<.001)	
Number of obs, 1547; LR chi2(39),	615.56; Prob > chi2, 0.00; Log I	ikelihood, -1582.65; Pseudo R2, 0.16

Variable	Public		Private		Informal Unemp		employed
	Coef.	ME	Coef.	ME	Coef.	ME	ME
Age	0.77***	0.03	0.19**	0.01	0.20***	0.02	-0.06
Age squared	-0.01***	0.00	0.00***	0.00	0.00***	0.00	0.00
Head	1.87***	0.04	1.54***	0.06	1.75***	0.28	-0.38
Household size	-0.15*	-0.01	-0.09	-0.01	-0.05	-0.01	0.02
Married	-0.02	0.04	-4.33***	-0.54	-1.39**	0.02	0.48
Primary	1.25	0.05	1.13*	0.11	0.24	-0.02	-0.13
Secondary	3.50***	0.19	1.61***	0.13	0.15	-0.11	-0.21
University	2.71**	0.22	2.37***	0.36	-0.66	-0.29	-0.29
Tenure	0.33	0.02	-0.15	-0.01	-0.06	-0.01	0.01
Ch0_6	-0.17	-0.01	0.11	0.01	0.11	0.02	-0.02
Ch7_17	0.36	0.01	0.16	-0.01	0.54**	0.11	-0.11
Relatives	0.69**	0.03	0.08	0.00	0.13	0.01	-0.04
Age Married	0.01	0.00	0.11***	0.01	0.04**	0.01	-0.01
Constant	-17.99***		-4.70***		-3.97***		

Table B8a: Females 15–34 age cohort without age-marital status interaction term

	Public		Private		Informal Ur		nemployed	
Variable	Coef.	ME	Coef.	ME	Coef.	ME	ME	
Age	1.14*	0.03	-0.05	-0.01	0.05	0.00	-0.02	
Age squared	-0.02	0.00	0.00	0.00	0.00	0.00	0.00	
Head	1.72***	0.03	1.39***	0.06	1.65***	0.28	-0.37	
Household size	-0.07	0.00	-0.08	-0.01	-0.06	-0.01	0.02	
Married	0.06	0.01	-1.64***	-0.18	-0.37	0.00	0.18	
Primary	0.71	0.01	1.24	0.09	0.95**	0.15	-0.25	
Secondary	3.21**	0.09	1.76**	0.14	0.76*	0.06	-0.29	
University	2.53	0.15	2.06*	0.36	-1.04	-0.26	-0.25	
Ownership of								
dwelling unit	0.70	0.03	-0.04	0.00	-0.23	-0.05	0.03	
Presence of								
children Age 0-6	6							
in a household	0.04	0.00	0.13	0.00	0.27	0.05	-0.05	
Presence of								
children age 7-	17							
in a household	0.04	0.00	0.22	0.00	0.51*	0.10	-0.10	
Presence of								
female relatives								
in a household	0.67*	0.02	0.11	0.01	0.08	0.01	-0.03	
Constant	-23.15***	-	-2.63	-	-3.20	-		

⁽Note: * p<.05; ** p<.01; *** p<.001)

Number of obs, 1090; LR chi2(36), 396.75; Prob > chi2, 0.00; Log likelihood, -1080.34; Pseudo R2, 0.16

Table B8b: Females 15–34 age cohort without age-marital status interaction term

(Note: * p<.05; ** p<.01; *** p<.001)

Number of obs, 1090; LR chi2(39), 414.09; Prob > chi2, 0.00; Log likelihood, 1071.67; Pseudo R2, 0.16

	Public		Private		Informal		Unemployed	
Variable	Coef.	ME	Coef.	ME	Coef.	ME	ME	
Age	1.28**	0.03	0.20	0.01	0.24	0.03	-0.07	
Age squared	-0.02*	0.00	0.00	0.00	0.00	0.00	0.00	
Head	1.93***	0.03	1.60***	0.06	1.91***	0.32	-0.42	
Household size	-0.07	0.00	-0.07	0.00	-0.06	-0.01	0.02	
Married	-4.49	-0.04	-6.06***	-0.46	-4.30***	-0.33	0.82	
Primary	0.76	0.01	1.24	0.09	0.98**	0.15	-0.25	
Secondary	3.23**	0.09	1.73**	0.13	0.75*	0.06	-0.28	
University	2.48	0.16	1.93	0.33	-1.13	-0.26	-0.23	
Tenure	0.73	0.03	0.00	0.00	-0.21	-0.05	0.02	
Ch0_6	0.00	0.00	0.08	0.00	0.23	0.05	-0.04	
Ch7_17	-0.11	-0.01	0.03	-0.01	0.35	0.08	-0.06	
Relatives	0.63*	0.02	0.07	0.00	0.04	0.00	-0.02	
Age married	0.18*	0.00	0.18**	0.01	0.16***	0.03	-0.04	
Constant	-23.64***		-4.68		-4.42*			

Table B9: Female

35-64 age-cohort (Note: * p<.05; ** p<.01; *** p<.001) Number of obs, 457; LR chi2(36), 196.81; Prob > chi2, 0.00; Log likelihood, -485.82; Pseudo R2, 0.17

Variable	Public		Private		Informal		Unemployed	
	Coef.	ME	Coef.	ME	Coef.	ME	ME	
Age	1.08	0.07	0.78*	0.06	0.04	-0.07	-0.07	
Age squared	-0.01*	0.00	-0.01*	0.00	0.00	0.00	0.00	
Head	2.0***	0.07	1.69***	0.05	1.79***	0.26	-0.37	
Household size	-0.24*	-0.02	-0.16	-0.01	-0.01	0.01	0.02	
Married	0.55	0.03	0.45	0.03	0.23	0.02	-0.07	
Primary	1.56	0.15	1.11	0.12	-0.37	-0.22	-0.05	
Secondary	3.74***	0.41	1.54*	0.09	-0.25	-0.33	-0.18	
University	2.79*	0.26	2.90**	0.39	-0.56	-0.40	-0.25	
Tenure	-0.02	0.00	-0.38	-0.04	0.15	0.06	-0.01	
Ch0_6	-0.82	-0.04	0.05	0.04	-0.64	-0.13	0.12	
Ch7_17	0.97*	0.04	0.39	-0.01	0.74	0.13	-0.16	
Relatives	0.75	0.06	-0.04	-0.02	0.16	0.01	-0.05	
Constant	-24.78*		-19.37*		-0.11			

Other recent publications in the AERC Research Papers Series:

Determinants of Private Investment Behaviour in Ghana, by Yaw Asante, Research Paper 100.

An Analysis of the Implementation and Stability of Nigerian Agricultural Policies, 1970–1993, by P. Kassey Garba, Research Paper 101.

Poverty, Growth and Inequality in Nigeria: A Case Study, by Ben E. Aigbokhan, Research Paper 102.

Effect of Export Earnings Fluctuations on Capital Formation, by Godwin Akpokodje, Research Paper 103.

- Nigeria: Towards an Optimal Macroeconomic Management of Public Capital, by Melvin D. Ayogu, Research Paper 104.
- International Stock Market Linkages in South Africa, by K.R. Jefferis, C.C. Okeahalam and T.T. Matome, Research Paper 105.

An Empirical Analysis of Interest Rate Spread in Kenya, by Rose W. Ngugi, Research Paper 106.

- *The Parallel Foreign Exchange Market and Macroeconomic Perfromance in Ethiopia*, by Derrese Degefa, Reseach Paper 107.
- Market Structure, Liberalization and Performance in the Malawi Banking Industry, by Ephraim W. Chirwa, Research Paper 108.

Liberalization of the Foreign Exchange Market in Kenya and the Short-Term Capital Flows Problem, by Njuguna S. Ndung'u, Research Paper 109.

External Aid Inflows and the Real Exchange Rate in Ghana, by Harry A. Sackey, Research Paper 110.

- Formal and Informal Intitutions' Lending Policies and Access to Credit by Small-Scale Enterprises in Kenya: An Empirical Assessment, by Rosemary Atieno, Research Paper 111.
- Financial Sector Reform, Macroeconomic Instability and the Order of Economic Liberalization: The Evidence from Nigeria, by Sylvanus I. Ikhinda and Abayomi A. Alawode, Research Paper 112.

The Second Economy and Tax Yield in Malawi, by C. Chipeta, Research Paper 113.

- Promoting Export Diversification in Cameroon: Toward Which Products? by Lydie T. Bamou, Research Paper 114. Asset Pricing and Information Efficiency of the Ghana Stock Market, by Kofi A. Osei, Research Paper 115.
- An Examination of the Sources of Economic Growth in Cameroon, by Aloysius Ajab Amin, Research Paper 116.
- *Trade Liberalization and Technology Acquisition in the Manufacturing Sector: Evidence from Nigeria*, by Ayonrinde Folasade, Research Paper 117.
- *Total Factor Productivity in Kenya: The Links with Trade Policy*, by Joseph Onjala, Research Paper 118. *Kenya Airways: A Case Study of Privatization*, by Samuel Oyieke, Research Paper 119.
- Determinants of Agricultural Exports: The Case of Cameroon, by Daniel Gbetnkon and Sunday A. Khan, Research Paper 120.
- Macroeconomic Modelling and Economic Policy Making: A Survey of Experiences in Africa, by Charles Soludo, Research Paper 121.
- *Determinants of Regional Poverty in Uganda*, by Francis Okurut, Jonathan Odwee and Asaf Adebua, Research Paper 122.
- *Exchange Rate Policy and the Parallel Market for Foreign Currency in Burundi*, by Janvier D. Nkurunziza, Research Paper 123.
- Structural Adjustment, Poverty and Economic Growth: An Analysis for Kenya, by Jane Kabubo-Mariara and Tabitha W. Kiriti, Research Paper 124.
- Liberalization and Implicit Government Finances in Sierra Leone, by Victor A.B. Davis, Research Paper 125.
- *Productivity, Market Structure and Trade Liberalization in Nigeria*, by Adeola F. Adenikinju and Louis N. Chete, Research Paper 126.
- Productivity Growth in Nigerian Manufacturing and Its Correlation to Trade Policy Regimes/Indexes (1962– 1985), by Louis N. Chete and Adeola F. Adenikinju, Research Paper 127.

Financial Liberalization and Its Implications for the Domestic Financial System: The Case of Uganda, by Louis A. Kasekende and Michael Atingi-Ego, Research Paper 128.

- Public Enterprise Reform in Nigeria: Evidence from the Telecommunications Industry, by Afeikhena Jerome, Research Paper 129.
- Food Security and Child Nutrition Status among Urban Poor Households in Uganda: Implications for Poverty Alleviation, by Sarah Nakabo-Sswanyana, Research Paper 130.
- Tax Reforms and Revenue Mobilization in Kenya, by Moses Kinyanjui Muriithi and Eliud Dismas Moyi, Research Paper 131.
- *Wage Determination and the Gender Wage Gap in Kenya: Any Evidence of Gender Discrimination?* by Jane Kabubo-Mariara, Research Paper 132.

Trade Reform and Efficiency in Cameroon's Manufacturing Industries, by Ousmanou Njikam, Research Paper 133. *Efficiency of Microenterprises in the Nigerian Economy*, by Igbekele A. Ajibefun and Adebiyi G Daramola, Research Paper 134.

- *The Impact of Foreign Aid on Public Expenditure: The Case of Kenya*, by James Njeru, Research Paper 135. Njikam, Research Paper 136.
- How Tied Aid Affects the Cost of Aid-Funded Projects in Ghana, by Barfour Osei, Research Paper 137.
- Exchange Rate Regimes and Inflation in Tanzania, by Longinus Rutasitara, Research Paper 138.
- Private Returns to Higher Education in Nigeria, by O.B.Okuwa, Research Paper 139.
- Uganda's Equilibrium Real Exchange Rate and Its Implications for Non-Traditional Export
- Performance, by Michael Atingi-Ego and Rachel Kaggwa Sebudde, Research Paper 140.
- Dynamic Inter-Links among the Exchange Rate, Price Level and Terms of Trade in a Managed Floating Exchange Rate System: The Case of Ghana, by Vijay K. Bhasin, Research Paper 141.
- Financial Deepening, Economic Growth and Development: Evidence from Selected Sub-Saharan African Countries, by John E. Udo Ndebbio, Research Paper 142.
- The Determinants of Inflation in South Africa: An Econometric Analysis, by Oludele A. Akinboade, Franz K. Siebrits and Elizabeth W. Niedermeier, Research Paper 143.
- The Cost of Aid Tying to Ghana, by Barfour Osei, Research Paper 144.
- A Positive and Normative Analysis of Bank Supervision in Nigeria, by A. Soyibo, S.O. Alashi and M.K. Ahmad, Research Paper 145.
- The Determinants of the Real Exchange Rate in Zambia, by Kombe O. Mungule, Research Paper 146.

An Evaluation of the Viability of a Single Monetary Zone in ECOWAS, by Olawale Ogunkola, Research Paper 147. Analysis of the Cost of Infrastructure Failures in a Developing Economy: The Case of Electricity Sector in Nigeria, by Adeola Adenikinju, Research Paper 148.

- *Corporate Governance Mechanisms and Firm Financial Performance in Nigeria*, by Ahmadu Sanda, Aminu S. Mikailu and Tukur Garba, Research Paper 149.
- Female Labour Force Participation in Ghana: The Effects of Education, by Harry A. Sackey, Research Paper 150.
- The Integration of Nigeria's Rural and Urban Foodstuffs Market, by Rosemary Okoh and P.C. Egbon, Research Paper 151.
- Determinants of Technical Efficiency Differentials amongst Small- and Medium-Scale Farmers in Uganda: A Case of Tobacco Growers, by Marios Obwona, Research Paper 152.
- Land Conservation in Kenya: The Role of Property Rights, by Jane Kabubo-Mariara, Research Paper 153.
- Technical Efficiency Differentials in Rice Production Technologies in Nigeria, by Olorunfemi Ogundele, and Victor Okoruwa, Research Paper 154.
- *The Determinants of Health Care Demand in Uganda: The Case Study of Lira District, Northern Uganda,* by Jonathan Odwee, Francis Okurut and Asaf Adebua, Research Paper 155.
- Incidence and Determinants of Child Labour in Nigeria: Implications for Poverty Alleviation, by Benjamin C. Okpukpara and Ngozi Odurukwe, Research Paper 156.
- *Female Participation in the Labour Market: The Case of the Informal Sector in Kenya*, by Rosemary Atieno, Research Paper 157.
- The Impact of Migrant Remittances on Household Welfare in Ghana, by Peter Quartey, Research Paper 158.
- Food Production in Zambia: The Impact of Selected Structural Adjustments Policies, by Muacinga C.H. Simatele, Research Paper 159.
- Poverty, Inequality and Welfare Effects of Trade Liberalization in Côte d'Ivoire: A Computable General Equilibrium Model Analysis, by Bédia F. Aka, Research Paper 160.
- *The Distribution of Expenditure Tax Burden before and after Tax Reform: The Case of Cameroon*, by Tabi Atemnkeng Johennes, Atabongawung Joseph Nju and Afeani Azia Theresia, Research Paper 161.
- Macroeconomic and Distributional Consequences of Energy Supply Shocks in Nigeria, by Adeola F. Adenikinju and Niyi Falobi, Research Paper 162.
- Analysis of Factors Affecting the Technical Efficiency of Arabica Coffee Producers in Cameroon, by Amadou Nchare, Research Paper 163.
- Fiscal Policy and Poverty Alleviation: Some Policy Options for Nigeria, by Benneth O. Obi, Research Paper 164. FDI and Economic Growth: Evidence from Nigeria, by Adeolu B. Ayanwale, Research Paper 165.
- An Econometric Analysis of Capital Flight from Nigeria: A Portfolio Approach, by Akanni Lawanson, Research Paper 166.

Extent and Determinnanttts of Child Labour in Uganda, by Tom Mwebaze, Research Paper 167.

Oil Wealth and Economic Growth in Oil Exporting African Countries, by Olomola Philip Akanni, Research

Paper 168.

- Implications of Rainfall Shocks for Household Income and Consumption in Uganda, by John Bosco Asiimwe, Research Paper 169.
- *Relative Price Variability and Inflation: Evidence from the Agricultural Sector in Nigeria*, by Obasi O. Ukoha, Research Paper 170.
- A Modelling of Ghana's Inflation: 1960–2003¹, by Mathew Kofi Ocran, Research Paper 171.
- *The Determinants of School and Attainment in Ghana: A Gender Perspective*, by Harry A. Sackey, Research Paper 172.
- Private Returns to Education in Ghana: Implications for Investments in Schooling and Migration, by Harry A. Sackey, Research Paper 173.
- Oil Wealth and Economic Growth in Oil Exporting African Countries, by Olomola Philip Akanni, Research Paper 174.
- *Private Investment Behaviour and Trade Policy Practice in Nigeria*, by Dipo T. Busari and Phillip C. Omoke, Research Paper 175.

Determinants of the Capital Structure of Ghanaian Firms, by Jochua Abor, Research Paper 176.

- *Privatization and Enterprise Performance in Nigeria: Case Study of some Privatized Enterprises*, by Afeikhena Jerome, Research Paper 177.
- Sources of Technical Efficiency among Smallholder Maize Farmers in Southern Malawi, by Ephraim W. Chirwa, Research Paper 178.
- Technical Efficiency of Farmers Growing Rice in Northern Ghana, by Seidu Al-hassan, Research Paper 179.
- Empirical Analysis of Tariff Line-Level Trade, Tariff Revenue and Welfare Effects of Reciprocity under an Economic Partnership Agreement with the EU: Evidence from Malawi and Tanzania, by Evious K. Zgovu and Josaphat P. Kweka, Research Paper 180.
- *Effect of Import Liberalization on Tariff Revenue in Ghana*, by William Gabriel Brafu-Insaidoo and Camara Kwasi Obeng, Research Paper 181.
- Distribution Impact of Public Spending in Cameroon: The Case of Health Care, by Bernadette Dia Kamgnia, Research Paper 182.
- Social Welfare and Demand for Health Care in the Urban Areas of Côte d'Ivoire, by Arsène Kouadio, Vincent Monsan and Mamadou Gbongue, Research Paper 183.
- *Modelling the Inflation Process in Nigeria*, by Olusanya E. Olubusoye and Rasheed Oyaromade, Research Paper 184.
- Determinants of Expected Poverty Among Rural Households in Nigeria, by O.A. Oni and S.A. Yusuf, Research Paper 185.
- Exchange Rate Volatility and Non-Traditional Exports Performance: Zambia, 1965–1999, by Anthony Musonda, Research Paper 186.
- Macroeconomic Fluctuations in the West African Monetary Union: A Dynamic Structural Factor Model Approach, by Romain Houssa, Research Paper 187.
- Price Reactions to Dividend Announcements on the Nigerian Stock Market, by Olatundun Janet Adelegan, Research Paper 188.
- Does Corporate Leadership Matter? Evidence from Nigeria, by Olatundun Janet Adelegan, Research Paper 189.
- Determinants of Child Labour and Schooling in the Native Cocoa Households of Côte d'Ivoire, by Guy Blaise Nkamleu, Research Paper 190.
- Analysis of Health Care Utilization in Côte d'Ivoire, by Alimatou CISSE, Research Paper 191.
- Poverty and the Anthropometric Status of Children: A Comparative Analysis of Rural and Urban Household in Togo, by Kodjo Abalo, Research Paper 192.
- African Economic and Monetary Union (WAEMU)¹, by Sandrine Kablan, Research Paper 193.